U.S. Customs and Border Protection

Slip Op. 15-15

SINCE HARDWARE (GUANGZHOU) CO., LTD., Plaintiff, v. UNITED STATES, Defendant, and Home Products International, Inc., Defendant-Intervenor.

> Before: Richard K. Eaton, Senior Judge Court No. 09–00123 PUBLIC VERSION

[Plaintiff's motion for judgment on the agency record is granted, and the matter is remanded to the Department of Commerce.]

Dated: February 18, 2015

William E. Perry, Dorsey & Whitney LLP, of Seattle, WA, argued for plaintiff. With him on the brief was *Emily Lawson*.

Carrie A. Dunsmore, Trial Attorney, Commercial Litigation Branch, Civil Division, U.S. Department of Justice, of Washington, D.C., argued for defendant. With her on the brief were Stuart F. Delery, Assistant Attorney General, Jeanne E. Davidson, Director, and Patricia M. McCarthy, Assistant Director. Of counsel on the brief was Nathaniel J. Halvorson, Attorney, Office of the Chief Counsel for Trade Enforcement and Compliance, U.S. Department of Commerce, of Washington, D.C.

Frederick L. Ikenson, Blank Rome LLP, of Washington, D.C., argued for defendantintervenor. With him on the brief was Larry Hampel.

OPINION AND ORDER

EATON, Senior Judge:

Before the court are the United States Department of Commerce's ("Commerce" or the "Department") final results, dated October 30, 2013, following remand of its antidumping review of *Floor-Standing*, *Metal-Top Ironing Tables and Certain Parts Thereof from the People's Republic of China*, 74 Fed. Reg. 11,085 (Dep't of Commerce Mar. 16, 2009) (final results of antidumping duty administrative review), and accompanying Issues and Decision Memorandum ("Issues & Dec. Mem.") (collectively, the "Final Results") made pursuant to the court's order in *Since Hardware (Guangzhou) Co. v. United States*, 37 CIT_,

__, Slip Op. 13–71 (2013) ("*Since Hardware III*"). *See* Final Results of Redetermination Pursuant to Ct. Remand (ECF Dkt. No. 169) ("Third Remand Results").

On this third remand,¹ Commerce was instructed to support with substantial evidence its claim that the import data from the United States Customs and Border Protection Agency (the "Customs Data") corroborated the 157.68-percent rate for plaintiff Since Hardware (Guangzhou) Co., Ltd.'s ("plaintiff" or "Since Hardware") subject merchandise. *See Since Hardware III*, 37 CIT at __, Slip Op. 13–71, at 16–17; Ex. 1 to Draft Second Results of Redetermination Pursuant to Ct. Remand (ECF Dkt. No. 179) ("Customs Data").

In the Third Remand Results, Commerce determined that its selected rate of 157.68 percent was corroborated, to the extent practicable, by information from independent sources. See Third Remand Results at 7. The Department then continued to assign the 157.68percent rate to Since Hardware's merchandise. See Third Remand Results at 7. Since Hardware objects to Commerce's determination, alleging that the Department failed to corroborate the assigned rate. See Pl. Since Hardware (Guangzhou) Co., Ltd.'s Objections to the Department of Commerce's Redetermination Pursuant to Third Remand 6 (ECF Dkt. No. 174) ("Pl.'s Br."). Defendant-intervenor, Home Products International, Inc., urges the court to sustain Commerce's determination. See Resp. of Home Products International, Inc., Def.int., to Pl.'s Objections to the Department of Commerce's Third Remand Determination 5 (ECF Dkt. No. 80). For the following reasons, the court finds that the Third Remand Results are not supported by substantial evidence and remands this matter to Commerce.

STANDARD OF REVIEW

"The court shall hold unlawful any determination, finding, or conclusion found . . . to be unsupported by substantial evidence on the record, or otherwise not in accordance with law." 19 U.S.C. § 1516a(b)(1)(B)(i). "The results of a redetermination pursuant to court remand are also reviewed for compliance with the court's remand

¹ The parties to this 2009 case have, no doubt, reached the point at which they wonder how much longer it can last, a point identified by a recent law review article:

The continual remands caused partly by the [Federal Circuit's] refusal to accept, or discouragement of, appeals following dispositive remands by the [United States Court of International Trade ("CIT")], together with lack of clear recognition of the authority of the CIT to issue a substantive reversal, can result in decisions that have no effect upon the parties and thus are purely advisory opinions. Under current practice, the CIT's decisions cease to be advisory and have some effect only when the administrative agency agrees to accept the CIT's direction, under protest, often after several remands, and the case proceeds to a final conclusive result. This is not a rational system for resolving cases in our constitutional system. This Article suggested various practical remedies, but the one principle that is clear is that Article III courts reviewing agency action for lawfulness have the power to reverse the decision of the reviewed agency.

Jane Restani & Ira Bloom, The Nippon Quagmire: Article III Courts and Finality of United States Court of International Trade Decisions, 39 BROOK. J. INT'L L. 1005, 1025 (2014).

order." Yantai Xinke Steel Structure Co. v. United States, 38 CIT __, __, Slip Op. 14–38, at 4 (2014) (quoting Xinjiamei Furniture (Zhangzhou) Co. v. United States, 38 CIT __, __, 968 F. Supp. 2d 1255, 1259 (2014)) (internal quotation marks omitted).

LEGAL FRAMEWORK

Pursuant to 19 U.S.C. § 1677e(a)(2)(C), the Department, when a party "significantly impedes a proceeding," must use "facts otherwise available . . . to fill in the gaps when Commerce has received less than the full and complete facts needed to make a determination' from the respondents." See 19 U.S.C. § 1677e(a)(2)(C); Foshan Shunde Yongjian Housewares & Hardware Co. v. United States, 35 CIT __, __, Slip Op. 11-123, at 7 (2011) (alteration in original) (quoting Gerber Food (Yunnan) Co. v. United States, 29 CIT 753, 767, 387 F. Supp. 2d 1270, 1283 (2005)) (internal quotation marks omitted). When Commerce has made the further finding that a party has failed to cooperate to the best of its ability, it "may use an inference that is adverse to the interests of that party in selecting from among the facts otherwise available," to determine that party's antidumping duty rate. 19 U.S.C. § 1677e(b). The use of this inference by Commerce is referred to as adverse facts available ("AFA").

When determining a rate based on an adverse inference, Commerce may use appropriate "secondary information." *See* Uruguay Round Agreements Act, Statement of Administrative Action, H.R. DOC.NO. 103–316, at 870 (1994), *reprinted in* 1994 U.S.C.C.A.N. 4040, 4199 ("SAA") ("Secondary information is information derived from the petition that gave rise to the investigation or review, the final determination concerning the subject merchandise, or any previous review under section 751 [(19 U.S.C. § 1675)] concerning the subject merchandise.").

Although the Department is authorized to use an adverse inference with respect to the facts used to determine rates for uncooperative parties, Congress has expressly established limitations on the Department when it relies on secondary information to assign a rate. *See* 19 U.S.C. § 1677e(c) ("When the [Department] . . . relies on secondary information rather than on information obtained in the course of an investigation or review, [Commerce] . . . shall, to the extent practicable, corroborate that information from independent sources that are reasonably at their disposal."). This corroboration requirement intentionally restricts the Department's ability to select questionable secondary information when it assigns a rate based on adverse inferences. *See* SAA, H.R. DOC.NO. 103–316, at 870, *re*- printed in 1994 U.S.C.C.A.N. at 4199; see also Hubscher Ribbon Corp. v. United States, 38 CIT __, __, Slip Op. 14–43, at 8 (2014) ("In practice 'corroboration' involves confirming that secondary information has 'probative value,' by examining its 'reliability and relevance." (quoting 19 C.F.R. § 351.308(d); Mittal Steel Galati S.A. v. United States, 31 CIT 730, 734, 491 F. Supp. 2d 1273, 1278 (2007))).

In order to satisfy the relevance requirement, the information used to corroborate a rate must bear some relationship to the respondent. That is, where a rate based on adverse inferences is derived from secondary information, Commerce must support the rate by demonstrating that the information "has some grounding in [the] commercial reality" of the respondent during the period of review ("POR"). See Gallant Ocean (Thai.) Co. v. United States, 602 F.3d 1319, 1324 (Fed. Cir. 2010). Therefore, "the assignment of a rate resulting from an adverse inference [must be] based on secondary information [that is] corroborated by evidence showing that the rate is 'reliable and relevant to the particular respondent." Foshan Shunde Yongjian Housewares & Hardware Co. v. United States, 38 CIT __, __, Slip Op. 14-69, at 9 (2014) (emphasis added) (quoting Foshan Shunde Yongjian Housewares & Hardware Co. v. United States, 37 CIT __, __, Slip Op. 13–47, at 8 (2013)); see also F.lli De Cecco di Filippo Fara S. Martino S.p.A. v. United States, 216 F.3d 1027, 1032 (Fed. Cir. 2000) ("It is clear from Congress's imposition of the corroboration requirement in 19 U.S.C. § 1677e(c) that it intended for an adverse facts available rate to be a reasonably accurate estimate of the respondent's actual rate, albeit with some built-in increase intended as a deterrent to non-compliance. Congress could not have intended for Commerce's discretion to include the ability to select unreasonably high rates with no relationship to the respondent's actual dumping margin."). In addition, unlike the remainder of the record, it is the responsibility of the Department to develop the corroborating information. See Foshan Shunde, 38 CIT at , Slip Op. 14-69, at 10 ("It is important to note that Congress placed the obligation to corroborate secondary information using independent sources on the Department, not on the interested parties who are normally responsible for generating the administrative record.").

DISCUSSION

I. Background

This case involves Since Hardware's challenge of the Department's final results of the third administrative review of the antidumping duty order on floor-standing, metal-top ironing tables and certain parts thereof from the People's Republic of China ("PRC") (the "Order") for the POR, August 1, 2006 through July 31, 2007. See Final Results, 74 Fed. Reg. at 11,085. In the Final Results, Commerce found the use of facts available appropriate because "Since Hardware . . . provided unreliable and incomplete documentation in support of its claimed purchases of market economy inputs," and thereby significantly impeded Commerce's investigation. See Issues & Dec. Mem. at cmt. 1 (citing 19 U.S.C. § 1677e(a)(2)(C)). As a result of Since Hardware's failure to cooperate, Commerce further applied AFA in selecting the facts relating to Since Hardware's independence from the PRC government as well as to those facts surrounding Since Hardware's cost and country of origin information. See Issues & Dec. Mem. at cmt. 1. This led the Department to determine that Since Hardware could not establish its independence from the PRC government, and as a result, the Department assigned Since Hardware the PRC-wide antidumping duty rate of 157.68 percent. See Issues & Dec. Mem. at cmt. 1.

The court questioned the use of an adverse inference with respect to the information relating to Since Hardware's entitlement to a separate rate, and remanded the question twice. See Since Hardware (Guangzhou) Co. v. United States, 34 CIT __, __, Slip Op. 10-108, at 22–23 (2010) ("Since Hardware I"); Since Hardware (Guangzhou) Co. v. United States, 35 CIT __, __, Slip Op. 11-146, at 29-30 (2011). The Department issued its Second Remand Results on November 29, 2011. Final Results of Redetermination Pursuant to Ct. Remand (ECF Dkt. No. 133) ("Second Remand Results"). In the Second Remand Results, Commerce found that Since Hardware was "entitled to a separate rate." See Second Remand Results at 5. Commerce then determined a separate rate of 157.68 percent for Since Hardware, based on the rate of a cooperative respondent in the investigation. See Second Remand Results at 7–8. Commerce supported its selection of that rate, in part, by explaining the importance of incentivizing future cooperation. See Second Remand Results at 9.

Commerce also offered several reasons why the rate was "both reliable and relevant," and therefore corroborated.² See Second Remand Results at 7. First, the Department asserted that the rate was reliable because it was calculated for a respondent during the investigation. Second Remand Results at 7–8. To corroborate the relevance of its selected rate to Since Hardware, Commerce used the Customs

² In *Since Hardware III*, the court found that the rate was reliable. *See Since Hardware III*, 37 CIT at __, Slip Op. 13–71, at 11 ("Plaintiff has neither placed evidence on the record to challenge the reliability of the selected rate, nor pointed to any court's holding declaring the selected rate to be unreliable. Thus, Commerce has sufficiently demonstrated reliability.").

Data derived from rates at which imports of the ironing tables, that entered into the United States during the POR from non-party producers and exporters, were eventually liquidated. *See* Second Remand Results at 8. In doing so, it argued that, because other companies were able to import and presumably sell ironing tables subject to the 157.68 percent rate, this meant that the rate was representative of the commercial reality of companies in the same business as plaintiff. *See* Second Remand Results at 8–9. In other words, Commerce contends that the liquidation rate for other exporters' entries of ironing tables, made during the POR, is indicative of their commercial reality, and thus that of Since Hardware, during that time period.

Commerce also expressly declined to reopen the record of the proceeding to gather more information and calculate a rate specific to Since Hardware. *See* Second Remand Results at 13. Plaintiff challenged Commerce's assigned rate of 157.68 percent, arguing that the rate did not reflect plaintiff's commercial reality during the POR. Second Remand Results at 11.

In Since Hardware III, the court rejected plaintiff's claim that Commerce was required to reopen the record, holding that "nothing in th[e] court's order directed that the Department must reopen the record." Since Hardware III, 37 CIT at __, Slip Op. 13–71, at 8 (citations omitted). Further, the court found that the rate of 157.68 percent was reliable, but found that Commerce did not demonstrate the relevance of that rate to plaintiff's commercial reality. See id. at __, Slip Op. 13–71, at 11. Accordingly, the court remanded the Second Remand Results. Id. at __, Slip Op. 13–71, at 16.

II. ANALYSIS

A. Commerce May Not Presume that the Highest Prior Margin Is Relevant

In this remand, Commerce endeavors to demonstrate that the assigned rate is relevant to plaintiff's commercial reality in several ways. First, it reasserts its position that the court should permit it to apply a modified version of the *Rhone Poulenc* presumption. *See* Third Remand Results at 10 ("[E]ven before looking to the corroborative evidence, the 157.68 percent rate is relevant to Since Hardware by virtue of its choice not to cooperate to the best of its ability." (citing SAA, H.R. DOC.NO. 103–316, at 870, *reprinted in* 1994 U.S.C.C.A.N. at 4199)). By using this presumption, Commerce attempts to assume that the highest previously calculated rate in the history of the antidumping duty order is automatically relevant to a party that has failed to cooperate with Commerce's investigation. "In other words, Since Hardware's actions demonstrate that it preferred to avoid providing to the Department the necessary information for calculating its true margin. Instead of providing that information, Since Hardware knowingly chose to risk incurring an AFA rate of 157.68 percent." Third Remand Results at 10. Thus, the Department initially seeks to corroborate an assigned rate by means of a presumption rather than by using direct evidence.

The *Rhone Poulenc* presumption, which was first found in the Federal Circuit's case, *Rhone Poulenc v. United States*, confirmed Commerce's use of the "best information available"³ to determine a rate in those instances where a respondent had refused to answer the questionnaires or where the answers were highly deficient. *See Rhone Poulenc, Inc. v. United States*, 899 F.2d 1185, 1190 (Fed. Cir. 1990). Put differently, "the case stands for the proposition that a respondent can be assumed to make a rational decision to either respond or not respond to Commerce's questionnaires, based on which choice will result in the lower rate." *Tianjin Mach. Imp. & Exp. Corp. v. United States*, 35 CIT __, __, 752 F. Supp. 2d 1336, 1347 (2011).

This Court, however, has continually rejected the expansion of *Rhone Poulenc*-type presumptions beyond what has been permitted by the Federal Circuit. *See, e.g., Tianjin Mach.*, 35 CIT at __, 752 F. Supp. 2d at 1348. This restriction of the inference reflects the Federal Circuit's reasoning that the use of a *Rhone Poulenc*-type presumption must be based on the "common sense inference" found in the case itself. *See Rhone Poulenc*, 899 F.2d at 1190 ("It reflects a common sense inference that the highest prior margin is the most probative evidence of current margins because, if it were not so, the importer, knowing of the ["best information available"] rule, would have produced *current* information showing the margin to be less." (internal quotation marks omitted)).

The presumption, though, is particularly ill-suited as a substitute for corroboration. As this Court noted in *Tianjin Machinery*, *Rhone Poulenc* was decided before the Uruguay Round Agreements Act, which, for the first time, required corroboration of AFA rates based on

³ The *Rhone Poulenc* presumption allowed Commerce to assign a previously established rate to a respondent that did not supply current information from which a rate could be calculated. Thus, the Department could assign a prior margin as the "best information available," where a party had refused to respond to the Department's questionnaires. *See Rhone Poulenc, Inc. v. United States,* 899 F.2d 1185, 1190 (Fed. Cir. 1990). The "best information available" standard was superseded by the "facts available" and "adverse facts available" provisions following the Uruguay Round Agreements Act in 1994. *See SAA, H.R. DOC.NO.* 103–316, at 869–70, *reprinted in* 1994 U.S.C.C.A.N. at 4198–99; *see also Shandong Mach. Imp. & Exp. Co. v. United States,* 34 CIT ___, ___ n.3, Slip Op. 10–88, at 3 n.3 (2010) (citing SAA, H.R. DOC.NO. 103–316, at 868, *reprinted in* 1994 U.S.C.C.A.N. at 4198; *Shandong Huarong Mach. Co. v. United States,* 30 CIT 1269, 1282 n.9, 435 F. Supp. 2d 1261, 1274 n.9 (2006)).

secondary information. Tianjin Mach., 35 CIT at __, 752 F. Supp. 2d at 1347-48 ("Congress directed Commerce to make additional findings in AFA cases." (citing 19 U.S.C. § 1677e(c)); see also Foshan Shunde, 38 CIT at n.6, Slip Op. 14–69, at 9 n.6 (noting that the corroboration requirement of 19 U.S.C. § 1677e(c) was added as part of the Uruguay Round Agreements Act (citing SAA, H.R. DOC.NO. 103-316, at 869-70, reprinted in 1994 U.S.C.C.A.N. at 4198-99)). Thus, while *Rhone Poulenc* may provide a rationale for assigning a particular rate when a respondent has failed to adequately answer Commerce's questionnaires, it says nothing about whether that rate represents a respondent's commercial reality.⁴ First, because corroboration was not a requirement when Rhone Poulenc was decided, the case's application to the corroboration of a rate is beyond the scope of the Federal Circuit's holding. Second, "common sense" does not support the idea that a respondent who has been found to be uncooperative can be presumed to know that there is a "rule" that the assignment of a rate previously calculated for another respondent in an earlier segment of the proceeding is self-corroborating. This is because there is no such rule and it is difficult to see how there could be one. That is, the manner in which a respondent decides to behave with respect to answering questionnaires simply tells Commerce nothing about the market conditions in which the respondent operates. Thus, this Court has held that, when a respondent fails to cooperate when answering Commerce's questionnaires, Commerce cannot "dispense with [the] corroboration requirement by employing the Rhone Poulenc presumption." Tianjin Mach., 35 CIT at __, 752 F. Supp. 2d at 1348. Indeed, the court has found unlawful the use of the presumption in Since Hardware III, and finds no reason to revisit its conclusion. See Since Hardware III, 37 CIT at n.4, Slip Op. 13–71, at 11 n.4.

Thus, the Department's reliance on the *Rhone Poulenc* presumption does nothing to satisfy its duty of corroboration. Therefore, the Department may not rely on this presumption to corroborate the 157.68-percent rate.

⁴ The recent case of *Dongtai Peak Honey Indus. Co. v. United States* does not conflict with this conclusion. *See Dongtai Peak Honey Indus. Co. v. United States*, Appeal No. 14–1479 (Fed. Cir. Jan. 30, 2015). In that case, the Federal Circuit cites *KYD, Inc. v. United States* for the proposition that a rate determined in a prior segment is reliable. *See id.* at 20–21 (citing *KYD, Inc. v. United States*, 607 F.3d 760, 766–77 (Fed. Cir. 2010)). The Court, however, did not authorize the use of the presumption as a substitute for actual evidence of a respondent's commercial reality.

B. Commerce Has Not Supported with Substantial Evidence its Conclusions Relating to Liquidation Rates

In addition, the Department has not supported with substantial evidence its conclusions drawn from the Customs Data. The Customs Data placed on the record by Commerce comprises information for imports of ironing tables into the United States during the POR. The summary spreadsheet, from which the Customs Data is derived, is made up of numerous columns, including a column that lists the antidumping duty order to which the merchandise is subject, the liquidation rate, the value of the shipment subject to the antidumping duty order, the quantity of merchandise liquidated at that particular rate, and a description of the primary items invoiced in that shipment. See Customs Data at Tab 1. The Department relies on this data to corroborate the 157.68-percent rate⁵ by arguing that the "AD DUTY" column in the Customs Data "shows that the entries at issue contained subject merchandise that was liquidated at the 157.68 percent rate." ⁶ See Third Remand Results at 18. As Commerce explains, the values in this column are equal to the "AD-VALUE" column multiplied by the "AD RATE" column. See Third Remand Results at 18; Customs Data at 25. From this, Commerce concludes that the values in the "AD DUTY BOND" column represent both the amount of duty deposit and the amount at which the merchandise was liquidated.⁷ See Third Remand Results at 11–12 ("The Customs [D]ata available in this review consist of a summary spreadsheet that lists, *inter alia*, . . . the antidumping duty liquidation rate (variable AD Duty) However, regarding the tariff classification for all of the entries that were liquidated at the 157.68

⁵ The Customs Data has thirty-two columns: "CASENBR," "CNTRYOR," "CNTRYEX," "ORGNAME," "EXPNAME," "ENTRYNO," "ENTRYDTE," "CREDTE," "LIQDTE," "EX-PDTE," "HTS," "DESCRIPTION," "TYPE," "TYPEDESC," "UOM," "MFRNBR," "MFR-NAME," "IMPORTER," "CONSIGNEE," "IMPNAME," "AD_DUTY_BOND," "AD_DE-POSIT_VALUE," "AD_QTY," "AD_VALUE," "AD_RATE," "DISTPOR," "DPNAME," "PORTLAD," "PLADCNTRY," "PLADNAME," "YEAR," and "MONTH." Commerce relies on the "AD_DUTY_BOND" column (column 21). See Customs Data at Tab 1. Since Hardware relies on the "AD_DEPOSIT_VALUE" column (column 22). See Customs Data at 25. The court has an obligation to review the data as a whole, and other columns appear to be relevant to whether the Customs Data supports Commerce's Determination. The court refers to the "AD_QTY" column (column 23) and the "AD_VALUE" column (column 24). See Customs Data at 25.

⁶ The "AD_DUTY" column, to which Commerce refers, appears to be an excerpt from the Customs Data containing the same data as the "AD_DUTY_BOND" column in the full exhibit. *See* Customs Data at 18.

⁷ "Liquidation means the final computation or ascertainment of duties on entries for consumption or drawback entries." 19 C.F.R. § 159.1; see also Shinyei Corp. of Am. v. United States, 524 F.3d 1274, 1276 (Fed. Cir. 2008).

percent rate, we note that the variable 'CASENBR' lists the ironing tables case number of 'A-570–888–000,' and that the column 'AD_Duty' specifically includes the antidumping duty that was collected on the shipment." (quoting Customs Data at 4, 18)). Assuming the "AD_DUTY_BOND" column represents both the amount of the deposit and the liquidated amount, the Department posits that the "AD_RATE" column would represent both the cash-deposit rate and the ultimate liquidation rate. *See* Third Remand Results at 12; Customs Data at 18, 25. In other words, Commerce infers from the Customs Data that importers actually paid cash deposits at the 157.68-percent rate during the POR and that the merchandise was eventually liquidated at that rate.

Since Hardware disputes that the Customs Data shows that the importers actually paid cash deposits at the 157.68-percent rate. For instance, Since Hardware points to the "AD_DEPOSIT_VALUE" column, which does not reflect that 157.68-percent deposits were made on most of the entries.⁸ See Pl.'s Br. 8; Customs Data at 25. More, importantly, plaintiff asserts that the deposit rate, not the ultimate liquidation rate, is what could be representative of the commercial reality of exporters during the POR. Pl.'s Br. 8. Plaintiff contends that, if the cash-deposit rate for the relied upon entries during the POR was not 157.68 percent, then doubt is cast on whether the rate of 157.68 percent is representative of plaintiff's commercial reality during the POR. See Pl.'s Br. 9.

The parties have drawn conflicting conclusions from the Customs Data, and it appears that none of them know with certainty whether a cash deposit was actually paid on any of the relevant entries, or, if paid, the amount of the cash-deposit rate assessed.⁹ During oral argument, both Commerce and plaintiff conceded that they do not know what the "AD_DEPOSIT_VALUE" column represents. When pressed on the issue, counsel for Commerce admitted that she did not know whether a cash deposit was actually paid. *See* Confidential Oral Arg. at 6:05–6:55, *Since Hardware (Guangzhou) Co. v. United States*, Court No. 09–00123 (June 10, 2014).

Because Commerce has chosen to rely on the Customs Data as its primary source for corroborating the secondary information, the Department has the burden of explaining how this data serves to cor-

⁸ The "AD_DEPOSIT_VALUE" column reflects [[]] as the amount deposited and paid for all but [[]]. See Customs Data at 37. Plaintiff argues that this indicates that a cash deposit was actually paid on only [[]], and that the cash-deposit rate for the other entries must have been [[]]. See Pl.'s Br. 9.

⁹ There were [[]] relevant entries relied upon by Commerce. Third Remand Results at 12.

roborate the assigned rate. See Foshan Shunde, 38 CIT at __, Slip Op. 14–69, at 10 ("It is important to note that Congress placed the obligation to corroborate secondary information using independent sources on the Department, not on the interested parties who are normally responsible for generating the administrative record."). By statute, each "determination by Commerce must include 'an explanation of the basis for its determination." See NMB Sing. Ltd. v. United States, 557 F.3d 1316, 1320 (Fed. Cir. 2009) (quoting 19 U.S.C. 1677f(i)(3)(A)). Thus, where, as here, "the Department has reached important conclusions that are not fully explained with reference to record evidence [and relevant statutes or regulations], remand is appropriate for Commerce to 'explain its rationale . . . such that a court may follow and review its line of analysis, its reasonable assumptions, and other relevant considerations." See Since Hardware III, 37 CIT at __, Slip Op. 13–71, at 15 (alteration in original) (quoting Clearon Corp. v. United States, 35 CIT __, __, Slip Op. 11-142, at 27-28 (2011)) (internal quotation marks omitted).

The Federal Circuit in *Gallant Ocean* held that, "[a]lthough Commerce has discretion in choosing from a list of secondary information to support its adverse inferences, Commerce must select secondary information that has some grounding in commercial reality." *Gallant Ocean*, 602 F.3d at 1324. Thus, in order to have probative value, Commerce must at least prove that its interpretation of the data is more than mere guesswork. Accordingly, should the Department continue to rely on the Customs Data to corroborate the rate assigned to plaintiff, it must (1) provide a full explanation of what the data represents and (2) supply substantial evidence tending to demonstrate the amounts deposited during the POR. Based on the evidence now before the court, however, the Department has failed to show that substantial evidence supports its conclusion that the Customs Data corroborates the 157.68-percent rate.

C. The Ultimate Liquidation Rate Is Not Indicative of Commercial Reality During the POR

The Department contends that the 157.68-percent rate is reflective of plaintiff's commercial reality because, according to Commerce, at liquidation, importers actually paid duties assessed at this rate on entries made during the POR.¹⁰ See Third Remand Results at 19. Thus, Commerce believes that the "data do establish that some importers paid antidumping duties of 157.68 percent" and, as a result,

¹⁰ According to the Department, [[]] of ironing tables came into the United States during the POR. *See* Third Remand Results at 18. The Department claims that all of these entries were ultimately liquidated at the 157.68-percent rate. *See* Third Remand Results at 18.

the rate is relevant to plaintiff because "other companies were able to participate in the U.S. Market while their imports were liquidated at a rate of 157.68 percent of the value of the entered merchandise." *See* Third Remand Results at 12, 13.

Plaintiff argues that "[t]he antidumping rate [at which] the entries of subject merchandise may ultimately be liquidated does not establish the commercial reality for a respondent during a review period." Pl.'s Br. 8. That is, according to plaintiff, it is the cash-deposit rate, which is assessed during the relevant period, and not the ultimate liquidation rate, which is assessed after the POR, that could be representative of an exporter's commercial reality during the POR. *See* Pl.'s Br. 8. Thus, for plaintiff, commercial reality, at least for other exporters, is demonstrated by the amount paid when the merchandise entered the United States, not the rate at which the merchandise was ultimately liquidated months or years later. Put another way, plaintiff argues that, the fact that the entries were ultimately liquidated at an antidumping duty rate of 157.68 percent does not offer any insight into plaintiff's commercial reality during the POR.

Plaintiff is correct that, regardless of whether the liquidation rate of 157.68 percent may have been eventually paid by importers, that alone is not indicative of plaintiff's commercial reality during the POR. As noted, the Department conceded at oral argument that it did not know whether the Customs Data contained information as to the cash-deposit rate, which would have been the rate at which Since Hardware or other exporters of subject merchandise would have willingly made sales into the United States. The subsequently imposed liquidation rates of entries imported during the POR does not necessarily show that exporters were able to export subject merchandise at a 157.68-percent rate.

This is because, during the POR, only the cash-deposit rates were known to the market participants. Therefore, when making a decision to import the ironing tables, an importer (who normally pays the duties) would only know how much was being paid as a cash deposit and, based on that amount, reach a judgment as to whether it could make a profit. The ultimate liquidation rate of merchandise is not determined until after the completion of the review of the antidumping duty order. Because liquidation rates are only known after the review is completed, neither a respondent nor any potential importer would have had that information at the time of its decision to participate in the U.S. market. *See Foshan Shunde*, 38 CIT at __, Slip Op. 14–69, at 17 ("At the time of importation only the cash-deposit rates were known. The ultimate liquidation rate of merchandise subject to a review of an antidumping duty order is unknown until after the

completion of the review itself. Thus, for Commerce to assert that [the respondent] or other exporters 'chose' to participate in the U.S. market knowing that its products were subject to a 157.68 percent rate simply assumes too much.").

Thus, despite the Department's arguments to the contrary, any liquidation rate that differs from the cash-deposit rate of imports subject to the Order sheds no light on whether exporters willingly participated in the U.S. market while being aware that they were likely to be assessed antidumping duties of 157.68 percent. As a result, these liquidation rates do not serve to corroborate the 157.68percent rate because they do not constitute substantial evidence of the commercial reality of plaintiff or other importers during the POR.

D. Commerce Failed to Support with Substantial Evidence its Reliance on the Small Quantity of the Entries

On remand, Commerce was directed to "either identify record evidence indicating that the Customs Data represents a relevant quantity of exports of the subject merchandise or reopen the record to place such additional evidence thereon." Since Hardware III, 37 CIT at ___, Slip Op. 13–71, at 17. Thus, Commerce was required to explain why its reliance on a small number of entries of limited value supplied it with the substantial evidence necessary to support its determination that the 157.68-percent rate was sufficiently corroborated. Id. at Slip Op. 13 71, at 16 ("[T]he Department shall explain why the Customs Data represents a sufficiently large number of entries to demonstrate the relevance of the selected rate or . . . otherwise corroborate its selected rate in a manner supported by substantial evidence and in accordance with law."). When Commerce relies on minimal data, courts have held that the Department must establish why that data is sufficient to be considered relevant. For example, in Gallant Ocean, the Federal Circuit held that, "[b]ecause Commerce did not identify any relationship between the small number of unusually high dumping transactions with [the plaintiff's] actual rate, those transactions cannot corroborate the . . . rate." Gallant Ocean, 602 F.3d at 1324. In this case, as in Gallant Ocean, if the Department is relying upon a small amount of data, from a small number of entries,¹¹ which substantially deviates from the rest of the data as a whole, it must explain why that minority of data merits reliance.

¹¹ Commerce relied upon a total of [[]] entries made during the POR, that were liquidated at the 157.68-percent rate, upon which [[]] different importers paid antidumping duties, to determine that this rate was "reflective of the commercial reality during the instant review period for these [[]] importers." See Third Remand Results at 12–13.

The Department argues that, "when corroborating a margin selected based on adverse inferences, Commerce is not required to establish that the selected AFA rate is paid by most, all[,] or a specific number of the importers. . . . Commerce is required to establish only that the secondary information has probative value." Def.'s Resp. to Comments on the Remand Redetermination 14 (ECF Dkt. No. 183) ("Def.'s Br."). If by this, Commerce means "some probative value" or "a little probative value," then it misstates its responsibility. Commerce's responsibility is to corroborate its use of secondary information with substantial evidence. See Consol. Edison Co. of N.Y. v. Nat'l Labor Relations Bd., 305 U.S. 197, 229 (1938) ("Substantial evidence is more than a mere scintilla. It means such relevant evidence as a reasonable mind might accept as adequate to support a conclusion." (citations omitted)). A small amount of data can be insufficiently probative of plaintiff's commercial reality. This does not mean that Commerce is required to provide a certain number of entries for that information to have probative value, but it is required to explain why the limited data in this case is sufficient to substantiate the Department's reliance thereon. See Yangzhou Bestpak Gifts & Crafts Co. v. United States, 716 F.3d 1370, 1380 (Fed. Cir. 2013) ("What could have been a coincidental correlation of the three data points is not enough to be substantial supporting evidence of economic reality."); Gallant Ocean, 602 F.3d at 1324 ("Because Commerce did not identify any relationship between the small number of unusually high dumping transactions with Gallant's actual rate, those transactions cannot corroborate the adjusted petition rate."). The Department has not done so here. Should the Department continue to rely upon the Customs Data on remand, it must provide a sufficient explanation as to why its small quantity amounts to substantial evidence.

E. Commerce Did Not Adequately Explain the Relevance of the Subject Merchandise that Entered at Lower Rates

On remand, the court also directed Commerce to explain "the significance, if any, of the subject merchandise being entered at rates below the selected rate." *Since Hardware III*, 37 CIT at __, Slip Op. 13–71, at 16. In other words, the Department was required to explain why the large quantity of entries that purportedly entered the United States at lower rates than 157.68 percent did not also provide evidence of the commercial reality of many market participants and thus give some indication of plaintiff's commercial reality.

The Department argues that the entries liquidated at lower rates do not reflect plaintiff's commercial reality because they were assigned rates rather than calculated rates.¹² Third Remand Results at 22 (explaining that the selected, lower "rate was not a calculated rate," and that, "[r]ather than a calculated rate, this [selected, lower] rate represents the separate rate assigned to [other respondents] during the [less-than-fair-value investigation], and is the weightedaverage rate for Since Hardware and" the 157.68-percent rate assigned to a cooperating party). Further, according to Commerce, Since Hardware's filing of unreliable and incomplete "information has precluded the Department from determining Since Hardware's actual margin or from considering whether Since Hardware's commercial reality differed from the commercial reality of those companies detailed in the Customs [D]ata." Third Remand Results at 22. That is, for Commerce, Since Hardware's filing of unreliable and incomplete documentation "rendered the calculation of a 'true' margin for Since Hardware impossible. Lacking information to determine Since Hardware's own commercial reality, Commerce [thus] used a calculated rate from the [less-than-fair-value] investigation as it [was] representative of the market for the subject merchandise." Third Remand Results at 22–23. Moreover, the Department insists that, because the lower, selected "rate is not a calculated rate and, because the rate must serve as a deterrent, selecting a rate that is indicative of commercial reality is not the only . . . consideration when selecting an AFA rate." Def.'s Br. 17 (citations omitted).

Plaintiff contends that the Department was unwarranted in concluding that the lower rates were irrelevant to Since Hardware merely because they were assigned rates, rather than calculated rates. See Pl.'s Br. 15 ("The Department's very conclusion in this proceeding is that the 157.68 percent rate is reflective of the commercial reality during the instant review because importers paid the rate and exporters participated in the market while their merchandise is assessed at the [157.68] percent rate. As directly provided in the Customs Data, the [separate rate] is also a rate paid by importers and a rate by which exporters participated in the market. The Department has not provided explanation for why its conclusion should somehow be different for a separate rate ... as opposed to a calculated rate. This is particularly the case here where the [separate] rate is based, in part, on Since Hardware's rate. This Court should find the Department's redetermination in this regard unsupported, and re-

¹² The "separate rate" ([[]]), which plaintiff urges as the rate that should have been assigned to Since Hardware in this review as AFA, was the highest rate at which entries of subject merchandise were liquidated during the POR, and is the weighted-average of the rates calculated for Since Hardware and Shunde Yongjian Housewares Co., Ltd., which were the two respondents in the less-than-fair-value investigation. This was also the rate assigned to other separate rate companies that were not individually investigated.

mand to the Department to consider the additional lower rates in the Customs Data." (footnote omitted)).

The Department defends its finding, claiming that the rates at which subject merchandise were liquidated, which are lower than 157.68 percent, were not appropriate as an AFA rate. Def.'s Br. 17, 18 ("[A]s explained by Commerce, the [rate urged by plaintiff, which is] ... the weighted-average rate of the two respondents in the less than fair value investigation, ... is not a calculated rate and, because the rate must serve as a deterrent, selecting a rate that is indicative of commercial reality is not the only ... consideration when selecting an AFA rate. ... Thus, while [the rate sought by plaintiff] was the rate assigned to separate rate respondents in the less than fair value investigation, [that] rate would provide an uncooperative respondent, like Since Hardware, with a lower margin than the highest previously published rate which it had the opportunity to rebut and does not provide proper inducement for cooperation with Commerce's proceedings." (citations omitted)).

The Department has not sufficiently explained why the lower rates are not relevant for purposes of corroboration. While it may be that the Department has provided some explanation for why it did not use the separate rates as the basis for assigning a rate to Since Hardware, it has not explained why the rates lack probative value for purposes of corroboration. Instead, Commerce asserts that the entries were assigned rates, rather than calculated rates, but does not explain why this difference makes the lower rates irrelevant. If the relevance of the 157.68-percent rate is, as the Department argues, established by the fact that market participants are paying the rate, why does merchandise entered at an assigned rate also give some indication of commercial reality? That is, what is important here is the entered rate-the rate at which importers are willing to participate in the U.S. market—i.e., whether they believe that, despite paying the rate, they can make a profit. The Department, however, has not sufficiently explained the significance of the subject merchandise entered at lower rates, as directed on remand. See Since Hardware III, 37 CIT at , Slip Op. 13–71, at 16. The Department was required to explain why the assigned rate represented Since Hardware's commercial reality. See id. at , Slip Op. 13–71, at 17. Should the Department continue to rely on the Customs Data on remand, it must fully explain why merchandise entered at rates less than 157.68 percent is not relevant to this inquiry.

F. The Department Has Not Satisfied its Burden of Corroboration

Commerce maintains that it has satisfied its burden to corroborate the rate to the extent practicable. Third Remand Results at 10 ("The question of relevance, then, is whether the 157.68 percent rate has any basis in reality that corresponds with the POR. As we explain below, our examination of the evidence here demonstrates that the limited independent information on the record of this proceeding supports the selected rate and, therefore, that rate is corroborated to the extent practicable."). The court is unpersuaded. Merely stating that the Department has not identified other sources that support its assessment is an inadequate explanation to support the conclusion that the Customs Data is the only information relevant to Since Hardware. "[T]he Department is not excused from its obligation to corroborate secondary information used to draw an adverse inference simply because the interested parties have placed no corroborative information on the record." Foshan Shunde, 38 CIT at __, Slip Op. 14–69, at 11 (citing Wash. Int'l Ins. Co. v. United States, 34 CIT, n.3, Slip Op. 10-16, at 5 n.3 (2010)). If, on remand, Commerce continues to maintain that there are no other independent sources reasonably at its disposal that could corroborate the assigned rate, Commerce "must describe the steps that it has taken so that a reviewing [c]ourt can determine if the Department's finding that corroboration was not practicable is supported by substantial evidence and in accordance with law." Foshan Shunde, 38 CIT at __, Slip Op. 14-69, at 19-20 (citing Toyota Motor Sales, U.S.A., Inc. v. United States, 22 CIT 643, 651, 15 F. Supp. 2d 872, 877-79 (1998)). Thus, given the Department's scant explanation in the Third Remand Results of the steps it took to corroborate the assigned rate of 157.68 percent, the court must conclude that Commerce did not satisfy its burden.

G. Plaintiff's Claim that the Rate Is Inconsistent with the Remedial Nature of the Statute Is Waived

Plaintiff claims that "the Department['s] decisions do not provide U.S. importers full due process of law because the Department['s] investigations and reviews are not conducted pursuant to the Administrative Procedure[] Act." Pl.'s Br. 16. Plaintiff appears to assert that the rate is inconsistent with the remedial nature of the statute. See Pl.'s Br. 16. Commerce responds that the Federal Circuit has already rejected a similar argument. Def.'s Br. 19 (citing KYD, Inc. v. United States, 607 F.3d 760, 768 (Fed. Cir. 2010)). In any event, because plaintiff raises this argument here for the first time in its papers,

after three remands, it has waived this argument. *See Yantai*, 38 CIT at ___, Slip Op. 14–38, at 18 ("Moreover, plaintiff waived its claim that Commerce used the wrong type of steel for the steel input by failing to raise it before the court prior to the remand. Because plaintiff did not raise this issue until after remand, the court's instructions necessarily did not direct the Department to reconsider its selection of the input itself.").

CONCLUSION AND ORDER

For the foregoing reasons, it is hereby

ORDERED that the Third Remand Results are remanded; it is further

ORDERED that, on remand, Commerce shall issue a redetermination that complies in all respects with this Opinion and Order, is based on determinations that are supported by substantial record evidence, and is in all respects in accordance with law; it is further

ORDERED that Commerce support the rate assigned to Since Hardware by demonstrating that the information has some grounding in the commercial reality of plaintiff during the POR; it is further

ORDERED that Commerce no longer rely on the presumption that, because the 157.68 percent rate had already been calculated in a prior segment of the proceeding at the time Since Hardware took the risk of providing unreliable, incomplete, and unusable data, it could have anticipated the assignment of that rate, to corroborate the assigned rate to Since Hardware; it is further

ORDERED that, should the Department continue to rely upon the Customs Data, the Department shall clarify all apparent inconsistencies in the data and conclusively establish the cash-deposit rate for the relevant entries; it is further

ORDERED that the Department shall explain with specificity why the cash-deposit rates for other market participants during the POR tend to corroborate the selected rate; it is further

ORDERED that, should the Department continue to rely upon the Customs Data, the Department shall explain with specificity why the Customs Data represents a sufficient quantity of exports of the subject merchandise to be relevant to Since Hardware; it is further

ORDERED that, should the Department continue to rely upon the Customs Data, the Department shall explain with specificity the significance, if any, of the subject merchandise being entered at rates below the selected rate; it is further

ORDERED that the Department may reopen the record to solicit any information it finds to be necessary to make its determination; it is further ORDERED that, if Commerce reopens the record, it may seek clarification and further information from the U.S. Customs and Border Protection Agency regarding the Customs Data and what the data represents; it is further

ORDERED that Commerce may select a new antidumping duty rate for Since Hardware and corroborate that rate; and it is further

ORDERED that the remand results shall be due on June 18, 2015; comments to the remand results shall be due thirty (30) days following filing of the remand results; and replies to such comments shall be due fifteen (15) days following filing of the comments.

Dated: February 18, 2015

New York, New York

/s/ Richard K. Eaton RICHARD K. EATON

Slip Op. 15–16

VINH HOAN CORPORATION, et al., PLAINTIFFS, and BINH AN SEAFOOD JOINT STOCK COMPANY, Plaintiff-Intervenor, v. UNITED STATES, Defendant, and CATFISH FARMERS OF AMERICA, et al., Defendant-Intervenors.

> Before: Claire R. Kelly, Judge Consol. Court No. 13–00156 Public Version

[Remanding Commerce's antidumping duty administrative review.]

Dated: February 19, 2015

Matthew Jon McConkey, Mayer Brown LLP, of Washington, DC, argued for Vinh Hoan Corporation.

Jonathan Michael Freed, Trade Pacific, PLLC, of Washington, DC, argued for Vinh Quang Fisheries Corporation.

Ned Herman Marshak, Grunfeld Desiderio Lebowitz Silverman & Klestadt, LLP, of New York, NY, argued for Vietnam Association of Seafood Exporters and Producers and Anvifish Joint Stock Company. With him on the brief were Andrew Brehm Schroth, Dharmendra Narain Choudhary, and Kavita Mohan.

John Joseph Kenkel, deKieffer & Horgan, PLLC, of Washington, DC, argued for Binh An Seafood Joint Stock Company.

Ryan Michael Majerus, Trial Attorney, Commercial Litigation Branch, Civil Division, U.S. Department of Justice, of Washington, DC, argued for defendant. With him on the brief were *Stuart F. Delery*, Assistant Attorney General, *Jeanne E. Davidson*, Director, and *Franklin E. White*, *Jr.*, Assistant Director. Of Counsel on the brief was *Devin Scott Sikes*, Attorney, Office of Chief Counsel for Trade Enforcement & Compliance, U.S. Department of Commerce, of Washington, DC.

Nazakhtar Nikakhtar, Akin, Gump, Strauss, Hauer & Feld, LLP, of Washington, DC, argued for Catfish Farmers of America, et al. With her on the brief were *Henry David Almond* and *Valerie A. Slater*.

OPINION AND ORDER

Kelly, Judge:

This consolidated action comes before the court on USCIT Rule 56.2 motions for judgment on the agency record, challenging the Department of Commerce's ("Department" or "Commerce") determination in *Certain Frozen Fish Fillets From the Socialist Republic of Vietnam*, 78 Fed. Reg. 17,350 (Dep't Commerce Mar. 21, 2013) (final results of antidumping duty administrative review and new shipper reviews) (*"Final Results"*), as amended, 78 Fed. Reg. 29,323 (Dep't Commerce May 20, 2013) (*"Amended Final Results"*); see also Issues and Decision Memorandum for Certain Frozen Fish Fillets from the Socialist Republic of Vietnam, A-552–801, (Mar. 13, 2013), available at http://enforcement.trade.gov/frn/summary/VIETNAM/2013–06550–1.pdf (last visited February 10, 2015) (*"Decision Memo"*).

Vinh Hoan Corporation ("Vinh Hoan") commenced this action pursuant to section 516A of the Tariff Act of 1930, as amended, 19 U.S.C. § 1516a (2012).¹ The court consolidated Vinh Hoan's challenge with actions filed by Vietnam Association of Seafood Exporters and Producers (collectively "VASEP"), Binh An Seafood Joint Stock Company ("Binh An"), Anvifish Joint Stock Company and Vinh Quang Fisheries Corporation ("Anvifish and Vinh Quang"), and Catfish Farmers of America, an association of processors and growers, and individual U.S. catfish processors, America's Catch, Alabama Catfish Inc. dba Harvest Select Catfish, Inc., Heartland Catfish Company, Magnolia Processing, Inc. dba Pride of Pond, and Simmons Farm Raised Catfish, Inc. (collectively "Catfish Farmers of America"). In addition to the above named parties' Rule 56.2 motions, Catfish Farmers of America filed a response, as a defendant-intervenor, in opposition to motions by VASEP, Vinh Hoan, Anvifish and Vinh Quang, and Binh An. See Def.-Intervenors' Resp. Opp'n Pl.'s, Consol. Pls.', and Pl.-Intervenor's Rule 56.2 Mot. J. Agency R., May 22, 2014, ECF No. 79 ("Catfish Farmers of America's Resp."). Vinh Hoan also filed a response, as a defendant-intervenor, in opposition to Catfish Farmers of America's motion. See Resp. Def.-Intervenor Vinh Hoan Corp. Opp'n Pls. Catfish Farmers of America, et al.'s Rule 56.2 Mot. J. Upon Agency R. Supp. Pl. VASEP's Rule 56.2 Mot. J. Upon Agency R., May 22, 2014, ECF No. 80 (Vinh Hoan's Resp.").

Given the length of this opinion, an outline of the issues discussed should be helpful to the reader. After a brief explanation of the background in the proceeding as well as the basis for jurisdiction and

 $^{^1}$ Further citations to the Tariff Act of 1930, as amended, are to the relevant provisions of Title 19 of the U.S. Code, 2012 edition.

the standard of review the court will begin with a discussion of the legal framework for primary surrogate country selection. Part II of the opinion considers Commerce's methodology in selecting a primary surrogate country as well as Commerce's analysis of nonfish factors of production, financial statements and whole fish in selecting a primary surrogate country. Part III considers Commerce's analysis of specific surrogate values. Part IV reviews Commerce's treatment of certain sales as consignment sales. Part V addresses the inclusion of sample transactions in Commerce's margin calculation. Part VI analyzes a challenge to Commerce's normal value calculation for Vin Hoan, specifically its refusal to adjust the ratio used to value Vin Hoan's FOPs.

BACKGROUND

On October 3, 2011, Commerce initiated this eighth antidumping ("AD") duty administrative review covering subject imports entered during the period of review ("POR"), August 1, 2010 through July 31, 2011. See Certain Frozen Fish Fillets From the Socialist Republic of Vietnam, 77 Fed. Reg. 56,180, 56,180 (Dep't Commerce Sept. 12, 2012) (preliminary results of the eighth antidumping duty administrative review and ninth new shipper reviews, partial rescission of review, and intent to revoke order in part) ("Preliminary Results"). Because Commerce treats Vietnam as a nonmarket economy ("NME") it selected a surrogate market economy country to value the factors of production of the subject imports. Id. at 56,183. During the preliminary results, Commerce chose Bangladesh as the primary surrogate country. Id. at 56,184. However, for the final results, Commerce chose Indonesia as the primary surrogate country. Final Results at 17,351.

Commerce selected Anvifish and Vinh Hoan as mandatory respondents, initially assigning them respective rates of 1.34 dollars per kilogram ("USD/kg") and 0.19 USD/kg. *Final Results* at 17,352–53. Further, in accordance with its practice, Commerce assigned the average of these rates, 0.77 USD/kg, to nonselected, separate rate respondents. Commerce also assigned a rate of 2.11 USD/kg to those who did not rebut the presumption of government control. *Id.* After receiving ministerial error allegations the Department revised the margin for Anvifish to 2.39 USD/kg and for the separate rate companies to 1.29 USD/kg. *Amended Final Results* at 29,324.

VASEP, Vinh Hoan, Anvifish and Vinh Quang, and Binh An challenge Commerce's selection of Indonesia as the primary surrogate country. *See* Mem. Law Supp. Pl.'s Rule 56.2 Mot. J. Upon Agency R. 8–61, Nov. 14, 2013, ECF No. 37 ("VASEP's Br."); Mem. Law Supp. Vinh Hoan Corp.'s Mot. J. Upon Agency R. 54–60, Nov. 14, 2013, ECF

No. 39 ("Vinh Hoan's Br."); Mem. P. & A. Supp. Anvifish Joint Stock Co. and Vinh Quang Fisheries Corp. J. Upon Agency R. 10–26, Nov. 14, 2013, ECF No. 42 ("Anvifish's & Vinh Quang's Br."); Mot. J. On Agency R. Under USCIT Rule 56.2 1-3, Nov. 14, 2013, ECF No. 40 ("Binh An's Mot."). Vinh Hoan also challenges Commerce's fish oil byproduct offset calculation, the values assigned to labor, two energy inputs and several by-products, the inclusion of "freight-in" in the selling, general, and administrative expense ratio, the inclusion of sample transactions in Vinh Hoan's margin calculation, the use of facts available on all of Vinh Hoan's consignment sales, and the inland freight and brokerage and handling calculation. Vinh Hoan's Br. 7–54. Vinh Hoan further claims that if it were not for Commerce's numerous errors, it would have been entitled to a de minimis margin which would have resulted in the revocation of the order as to Vin Hoan. Vinh Hoan's Br. 60. Anvifish and Vinh Quang challenge Commerce's use of the Vitarich price quote to value Anvifish's byproducts and use of a Philippine company's financial statement to calculate financial ratios. Avifish's & Vinh Quang's Br. 19-33.² Binh An joins the arguments briefed by VASEP, Vinh Hoan, and Anvifish and Vinh Quang in all respects. Binh An's Mot. 1.

Catfish Farmers of America challenge Commerce's refusal to make an adjustment to Vinh Hoan's normal value ("NV") and Commerce's calculation of Vinh Hoan's fish oil byproduct offset. Mem. Supp. Pls. Catfish Farmers of America, et al.'s Rule 56.2 Mot. J. Upon Agency R. 6–25, Nov. 14, 2013, ECF No. 44 ("Catfish Farmers of America's Br.").

Defendant, United States ("Defendant"), responds that the court should sustain Commerce's determinations except for Commerce's calculation of Vinh Hoan's fish oil byproduct offset for which Defendant requests a voluntary remand. Def.'s Resp. Pls.' Mots. J. Upon Agency R. 2–3, May 22, 2014, ECF No. 77 ("Def.'s Resp.").

The court finds that Commerce's primary surrogate selection is contrary to law and unsupported by substantial evidence. Further, the court grants the Defendant's request for a voluntary remand to reconsider its fish oil byproduct calculation. The court also finds Commerce's use of facts available for all of Vinh Hoan's CCEP sales is unsupported by substantial evidence. Moreover, the court sustains Commerce's inclusion of sample sales in Vinh Hoan's margin calculation. Finally, the court finds that Commerce's refusal to adjust Vinh

² Anvifish and Vinh Quang submitted their USCIT R. 56.2 motion and accompanying memorandum on November 14, 2013. Subsequently, there was a substitution of attorney for Anvifish on May 8, 2014. For ease of reference the court will continue to refer to the memorandum filed November 14, 2013 as "Anvifish's & Vinh Quang's Br." and will refer to Anvifish and Vinh Quang together when discussing the arguments made in that brief.

Hoan's NV was not supported by substantial evidence. The court reserves judgment on the remaining issues.

JURISDICTION AND STANDARD OF REVIEW

The court has jurisdiction pursuant to 28 U.S.C. § 1581(c)(2012), and 19 U.S.C. § 1516a(a) and "shall hold unlawful any determination, finding or conclusion found . . . to be unsupported by substantial evidence on the record, or otherwise not in accordance with law" 19 U.S.C. § 1516a(b)(1)(B)(i).

DISCUSSION

I. Primary Surrogate Country Selection: Legal Framework and Background

In NME AD proceedings, Commerce generally calculates NV "on the basis of the value of the factors of production utilized in producing the merchandise . . . [together with] an amount for general expenses and profit plus the cost of containers, coverings, and other expenses." 19 U.S.C. § 1677b(c)(1). Commerce shall value the factors of production "based on the best available information regarding the values of such factors in a market economy country or countries considered to be appropriate by the administering authority." Id. Moreover, to the extent possible, Commerce shall use "the prices or costs of factors of production in one or more market economy countries that are-(A) at a level of economic development comparable to that of the nonmarket economy country, and (B) significant producers of comparable merchandise." 19 U.S.C. § 1677b(c)(4). While Commerce has broad discretion in deciding what constitutes the best available information, see QVD Food Co. v. United States, 658 F.3d 1318, 1323 (Fed. Cir. 2011) (noting the absence of a definition for "best available information" in the AD statute), it must ground its selection of the best available information in the overall purpose of the AD statute, calculating accurate dumping margins. See CS Wind Vietnam Co. v. United States, 38 CIT ____, ___, 971 F. Supp. 2d 1271, 1277 (2014) (citing Rhone Poulenc, Inc. v. United States, 899 F.2d 1185, 1191 (Fed. Cir. 1990)); see also Parkdale Int'l. v. United States, 475 F.3d 1375, 1380 (Fed. Cir. 2007).

Commerce's regulatory preference is to "value all factors in a single surrogate country." 19 C.F.R. § 351.408(c)(2)(2012).³ To implement this preference, Commerce selects a primary surrogate country, using

 $^{^3}$ Further citations to the Code of Federal Regulations is to the 2012 edition, unless otherwise noted.

a four step process. See Import Admin., U.S. Dep't Commerce, Non-Market Economy Surrogate Country Selection Process, Policy Bulletin 04.1 (2004), available at, http://enforcement.trade.gov/policy/ bull04–1.html (last visited February 11, 2014) ("Policy Bulletin 04.1"). The process follows sequentially: (1) the Office of Policy ("OP") assembles "a list of potential surrogate countries that are at a comparable level of economic development to the NME country"; (2) Commerce identifies countries from the list "with producers of comparable merchandise"; (3) Commerce "determines whether any of the countries which produce comparable merchandise are 'significant' producers of that comparable merchandise"; and (4) if more than one country satisfies steps (1)-(3), Commerce will select "the country with the best factors data." See generally id.

In this review, Commerce placed its list of potential surrogate countries on the record on November 22, 2011. See Antidumping Duty Administrative Review of Certain Frozen Fish Fillets from the Socialist Republic of Vietnam ("Vietnam"): Surrogate Country List, PD 22 at bar code 3042499–01 (Nov. 22, 2011) ("OP List").⁴ The OP List provided Per Capita GNI values for the year 2010 from the World Bank's World Development Report 2012 for Vietnam and six other countries that were deemed "economically comparable." Id. These GNI values were Vietnam \$1,100, Bangladesh \$640, Pakistan \$1,050, Nicaragua \$1,080, India \$1,340, Philippines \$2,050, and Indonesia \$2,580. Id. In the preliminary results, Commerce explained that its longstanding practice is "to identify those countries which are at a level of economic development similar to Vietnam in terms of gross national income ("GNI") data available in the World Development Report provided by the World Bank." *Preliminary Results* at 56,183.⁵ Further, with respect to the countries on the list, the Department stated that it was "satisfied that they [were] equally comparable in

 $^{^4}$ Defendant submitted an appendix to the administrative record, which can be found at ECF No. 27–1, on June 19, 2013.

⁵ Commerce's regulations provide that Commerce "place[s] primary emphasis on per capita GDP as the measure of economic comparability." 19 C.F.R. § 351.408(b). In 2007, Commerce amended its methodology choosing to rely on per capita GNI as opposed to per capita GDP "because while the two measures are very similar, per capita GNI is reported across almost all countries by an authoritative source (the World Bank), and because the Department believes that the per capita GNI represents the single best measure of a country's level of total income and thus level of economic development." Antidumping Methodologies in Proceedings Involving Non-Market Economy Countries: Surrogate Country Selection and Separate Rates, 72 Fed. Reg. 13246, 13246 n.2 (Dep't Commerce Mar. 21, 2007) (request for comment); see also Clearon Corp. v. United States, Slip Op. 14–88, 2014 WL 3643332, at *9–10 (CIT July 24, 2014) (finding Commerce's reliance on GNI reasonable and in accordance with law). The court notes that no party has raised the issue of whether using GNI to determine economic comparability is contrary to Commerce's regulation.

terms of economic development" *Id.*; see also Policy Bulletin 04.1 at n.5 (explaining that Commerce treats countries on the OP list as equal in terms of economic comparability because "the statute does not require the Department to use a surrogate country that is at a level of economic development most comparable to the NME country"). Next, Commerce found all six countries to be significant producers of comparable merchandise based on 2009 export data for frozen fish fillets from the United Nations Food and Agriculture Organization. Preliminary Results at 56,183. Because all six countries were economically comparable and significant producers of comparable merchandise, Commerce turned to data considerations. Id. Commerce explained that Bangladeshi Department of Agriculture Marketing, Ministry of Agriculture, online pangas price data ("DAM data") was the best available information on the record to value whole live fish, the factor of production which "account[ed] for the largest percentage of the NV as fish fillets are produced directly from whole live fish." Id. Based on this finding, Commerce selected Bangladesh as the primary surrogate country. Id. at 56,184. Commerce issued its preliminary results on September 12, 2012. After Commerce issued the preliminary results, the parties and Commerce continued to revisit the surrogate country selection.

On November 8, 2012, Commerce extended the deadline for submitting "surrogate value comments" to November 20, 2012. Extension of Surrogate Value Submissions, PD 284 at bar code 3105147–01 (Nov. 8, 2012). On November 20, 2012, VASEP submitted potential primary surrogate country lists for the administrative review and new shipper reviews covering the POR August 1, 2011 to July 31, 2012 ("Subsequent Review GNI Lists")⁶ in addition to GNI data for all countries from 2011 ("2011 GNI data") and GDP data for all countries from 2010 ("2010 GDP data").⁷ The petitioner below objected to VASEP's submission of the Subsequent Review GNI Lists "because they [did] not contain information to value respondents' factors of production." Catfish Farmers of America's Surrogate Value Rebuttal Data Cover Letter 1–2, PD 376 at bar code 3108782–01 (Dec. 4, 2012). VASEP responded by letter arguing that the exhibits should not be

⁶ Commerce rejected the Subsequent Review GNI Lists as untimely filed new factual information. They can be found as attachments to VASEP's brief. *See* App. Pl.'s Mem. Law Supp. Rule 56.2 Mot. J. Upon Agency R. Docs. 5–6, Nov. 21, 2013, ECF No. 51–1.

⁷ On the same day, Catfish Farmers of America submitted data to value the whole live fish input, which was accepted by Commerce. CFA Surrogate Value Submission Ex. 34, PD 287 at bar code 3106818–01 (Nov. 20, 2012) ("IAS Data"). As will be discussed elsewhere, the IAS data played an important role in Commerce's decision to select Indonesia rather than Bangladesh as the primary surrogate country.

rejected. VASEP's Response to Petitioners Post-Prelim Surrogate Value Rebuttal, PD 388 at bar code 3110023–01 (Dec. 11, 2012). The Department agreed with the petitioner. *See* Rejection of New Factual Information, PD 395 at bar code 3111617–01 (Dec. 21, 2012). VASEP resubmitted its submission excluding the Subsequent Review GNI Lists but including the 2011 GNI and 2010 GDP data. *See* VASEP's Resubmission of Surrogate Value Submission of November 20, 2012, PD 404 at bar code 3112136–01 (Dec. 26, 2012).

After the parties filed case briefs, Commerce issued its final results choosing Indonesia as the primary surrogate country. Commerce began its analysis by explaining that it was "satisfied that [all countries on the OP List were] equally comparable in terms of economic development," and finding that all countries on the OP List were "exporters of frozen fish fillets and, thus, significant producers of comparable merchandise." *Decision Memo* 3–10. Commerce then turned to data considerations to make its primary surrogate country selection. Commerce eliminated India, Nicaragua, and Pakistan, explaining that no party argued for valuing whole fish from those countries and, in any event, the record did not contain suitable information from those countries. *Id.* at 11. Next, Commerce set out to compare the data for whole fish values from Bangladesh, Indonesia, and the Philippines. At the outset, Commerce noted that it was

in the unusual situation of having on the record three sources of information issued by governments, which represent official statements of those governments as to the price of whole live fish -i.e., the *Philippines FS*, *Indonesian AS*, and *DAM Data* sources, and one source from an international organization, relevant to our analysis -i.e., the *FAO FIGIS Data* source. While we typically do not scrutinize official government statistics in such detail, the necessity to respond to the comments raised by interested parties and to select one of the sources compelled us to do so in this case.

Decision Memo 12 (footnote omitted).

In evaluating these sources, Commerce explained that it "considers several criteria, including whether the SV data is contemporaneous, publicly available, tax and duty exclusive, representative of a broad market average, and specific to the input." *Decision Memo* 19 (citation omitted). Commerce found all sources to be contemporaneous, publicly available, and tax and duty and exclusive. *Id.* at 19–20.

In its broad market average discussion, Commerce found all three sources to be broad market averages. *Id.* at 20–22 (disagreeing with respondents' contentions regarding Philippines FS and FAO FIGIS

Data). However, Commerce noted concerns with Philippines FS Data collection methods and found that it was a less robust data source than the other three. *Id.* at 20–21. Relying on this finding, Commerce found Philippines FS not to be the best available information on the record. Additionally, Commerce explained several differences between the *DAM Data* on the record of the current and prior administrative reviews. *Id.* 21–22. Commerce relied on these differences to find "*DAM Data* to not represent as broad a market average as it did in the last administrative review and not as broad a market average as the *Indonesian AS.*" *Id.* at 22.

Next, Commerce discussed specificity. After a lengthy discussion, Commerce found "that the FAO FIGIS Data is not as species specific as the DAM Data or Indonesian AS." Decision Memo 25. It explained that record evidence showed five species of pangasius were grown in Indonesia, with two species being commonly grown. Id. It was unclear which of these species were included in the FAO FIGIS Data. Id. Additionally, unlike Indonesian AS, FAO FIGIS Data did not list the types of aquaculture areas from which its data was collected.⁸ Id. This omission was relevant because the record showed that in Indonesia hypophthalmus (subject merchandise) was grown in ponds and cages whereas jambal (not subject merchandise) was grown in rivers. Id. at 24. Thus, using Indonesian AS, as opposed to FAO FIGIS Data, allowed Commerce to eliminate at least some of its concerns about inclusion of other species grown in Indonesia. Id. Relying on these findings and the fact that Indonesian AS was a primary source while FAO FIGIS Data was not, Commerce concluded that FAO FIGIS Data was not the best available information. Id. at 25. Commerce also expressed concerns about whether dead fish were included in DAM Data values. Commerce stated that respondents all indicated they only consume live fish. Id. at 23. Commerce further explained that "because dead fish sell for less than live fish, [it] harbor[ed] concerns that the DAM Data may understate the price of the whole live fish FOP." Id. at 23. Relying on these findings, Commerce did "not find the DAM Data to be as specific as . . ." Indonesian AS. Id.

Next, Commerce considered reliability. First, it noted "that the *DAM Data* and *Indonesian AS* both have price fluctuations." *Decision Memo* 26. It explained that it did not consider any of the "SV choices . . . to be anomalous with regard to price variances and, thus, consider[ed] all sources equal in this regard." *Id.* However, Commerce found differences between the weekly and yearly average prices in the DAM

 $^{^{\}rm 8}$ Aquacultures areas refer to different methods used to harvest the whole live fish input used to produce subject merchandise.

Data. As a result, Commerce "requested information from DAM concerning its collection and collation methods," to which DAM did not respond. *Id.* at 26. Commerce explained that it did not have the same concerns with regard to the Indonesian AS. *Id.* Concluding, Commerce explained Indonesian AS was the best available information on the record to value whole live fish because it did "not give rise to the unanswered questions posed by the extent to which dead fish is represented in the *DAM Data*, the extent to, and process by, which the *DAM Data* is examined for errors." *Decision Memo* 27.

Turning to considerations other than whole live fish, Commerce explained that "factors other than whole fish and surrogate ratios account for a significant portion of direct materials and NV" Decision Memo 10. This conclusion was based on findings that Vinh Hoan was substantially integrated, many Vietnamese exporters of subject merchandise had similar production experiences to Vinh Hoan, and Vinh Hoan was one of the largest exporters of subject merchandise. See id. at 9. It further explained that, "the far greater contemporaneity and demonstrated importance to the NV calculation of the additional FOP [sic] available from Indonesia" supported its primary surrogate country selection. Decision Memo 27; see also id. at 10 (finding that with two exceptions all surrogate values for Indonesia were more contemporaneous than their Bangladeshi counterparts). Finally, Commerce explained that it was selecting Indonesia as the primary surrogate country because it was economically comparable to Vietnam, a significant producer of comparable merchandise, provided the best available information to value whole live fish, and provided surrogate values for other factors of production that were much more contemporaneous than data from other potential surrogate countries. Id. at 27.

II. Primary Surrogate Country Selection: Analysis

Commerce's primary surrogate country selection suffered from several deficiencies and must be remanded for reconsideration for the following reasons. Commerce's refusal to consider 2011 GNI data that it had accepted as record evidence, refusal to consider relative economic comparability of potential surrogate countries, reliance on contemporaneity of nonfish factors to the exclusion of other data considerations, and failure to compare the relative quality of financial statements made its determination contrary to law. Additionally, these legal deficiencies caused Commerce to ignore relevant detracting evidence. Moreover, Commerce's preference for Indonesian data to value whole live fish was not supported by substantial evidence. Thus, Commerce's primary surrogate country selection is both contrary to law and not supported by substantial evidence.

A. Primary Surrogate Country Selection: Data and Methodology

1. Commerce Must Consider 2011 GNI Record Evidence

The parties dispute what evidence Commerce must consider in making its primary surrogate country selection. Commerce disregarded record evidence of 2011 GNI data, VASEP's Resubmission of Surrogate Value Submission of November 20, 2012 Ex. 3C, PD 404-405 at bar code 3112136-01-02, (Dec. 26, 2012), and 2010 GDP data, VASEP's First Surrogate Value Rebuttal Submission Ex. 20F, PD 212 at bar code 3081393-03 (June 18, 2012). See also VASEP's Br. at 16. It also rejected submission of the Subsequent Review GNI Lists as untimely filed "new factual information." Rejection of New Factual Information, PD 395 at bar code 3111617-01 (Dec. 21, 2012). VASEP argues Commerce's decision to consider the 2010 GNI data from the OP List, but not to consider 2011 GNI data and 2010 GDP data, and to reject Subsequent Review GNI Lists was contrary to law. The court finds that Commerce's decision to disregard the 2011 GNI data is contrary to law. However, the court sustains Commerce's decisions to disregard the 2010 GDP data and reject Subsequent Review GNI Lists.

Commerce set "November 22, 2011 through May 23, 2012" as the time for submitting "any information the Department should consider when selecting the surrogate country." Decision Memo 4 (citing Commerce Letter Re: Deadline for Surrogate Country and Surrogate Value Submissions, PD 127 at bar code 3075253-01 (Apr. 27, 2012)). VASEP submitted new information on November 20, 2012. Commerce accepted the GNI data and GDP data onto the record, but rejected the Subsequent Review GNI Lists as untimely filed new factual information pursuant to 19 C.F.R. § 351.301(b)(2). Despite accepting the 2011 GNI and 2010 GDP and putting it on the record, Commerce nevertheless refused to rely on that data when choosing a primary surrogate country explaining that it "considers the selection of potential surrogate countries to be similar to the selection of mandatory respondents in an administrative review . . . [it] must be resolved early in the case in order to provide sufficient time for party participation and the necessary analysis." Decision Memo 4. Commerce explained that the 2011 GNI and 2010 GDP data "was not available on the record of this review for the Department to use at the time we made our surrogate country selection." ⁹ Further, no party challenged the list of potential surrogate countries during that time. Finally, Commerce explained that, "[r]evising the list of surrogate countries at a later date would be potentially unfair to the parties and create undue administrative difficulties." *Decision Memo* 5.

The court sustains Commerce's decision to disregard the 2010 GDP data. Commerce stated that its long-standing practice has been to evaluate "GNI data available in the World Development Report provided by the World Bank." Decision Memo 3. See also Policy Bulletin 04.1 at n.4. Defendant explains that Commerce's decision not to consider GDP data was based on this past practice and apparent judicial affirmation. Def.'s Resp. 21-22. Additionally, Defendant argues that GDP data is unreliable because it is derived from two sources, the World Bank and the International Monetary Fund, and there is nothing on the record showing these organizations reported GDP on a consistent basis with each other. Id. at 22. VASEP has not attempted to distinguish its arguments between Commerce's disregard of the GDP data and its disregard of the GNI data. Thus, there is no challenge to Commerce's preference and practice of relying on GNI data to make its economically comparable determination before the court.¹⁰ Thus, the court finds that Commerce has provided a reasonable explanation for disregarding 2010 GDP data.

Commerce's rejection of the Subsequent Review GNI Lists was in accordance with law. The Department's regulations specify deadlines for when parties must make submissions. *See e.g.* 19 C.F.R. § 351.301(a) (providing that "[t]his section sets forth the time limits for submitting such factual information, including . . . publicly available information to value factors in nonmarket economy cases"). In general for final results of administrative reviews, factual information is due "140 days after the last day of the anniversary month" 19 C.F.R. § 351.301(b)(2). "Factual information" means "(i) [i]nitial and supplemental questionnaire responses; (ii) [d]ata or statements of

⁹ It is not clear what Commerce is referring to here when it says "at the time we made our surrogate country selection." Commerce made its surrogate country selection, choosing Indonesia, when it issued its final results on March 21, 2013. Clearly this information was available at this time. In the very next sentence Commerce states "[t]hus the Surrogate Country List which the Department released on November 22, 2011 contained the most up-to-date information . . ." *Decision Memo* 4. This sentence suggests that Commerce was referring to the date on which it issued the OP List, November 22, 2011. If Commerce is referring to the November 22, 2011 date, this position would appear to undercut the value of the comment period.

¹⁰ The court notes that Commerce's practice has been judicially affirmed in other cases, *see e.g. Jiaxing Bro. Fastener Co., v. United States*, 38 CIT ___, 961 F. Supp. 2d 1323 (2014); *Clearon Corp.*, Slip Op. 14–88, 2014 WL 3643332, but expresses no opinion as to the reasonableness of Commerce's methodology.

fact in support of allegations; (iii) [o]ther data or statements of facts; and (iv) [d]ocumentary evidence." 19 C.F.R. § 351.102(b)(21). In addition, notwithstanding 19 C.F.R. § 351.301(b)(2), interested parties may submit publicly available information to value factors of production within "20 days after the date of publication of the preliminary results of review." 19 C.F.R. § 351.301(c)(3)(ii).

Commerce rejected VASEP's submission of the Subsequent Review GNI Lists explaining that,

[t]hough styled as a submission filed pursuant to section 351.301(c)(3) of the Department's regulations, we find that Vinh Hoan Corporation's and VASEP's submissions of November 20, 2012 do not only contain information to value respondents' factors of production. Instead, these submissions contain new factual information pertaining to the selection of a surrogate country, and this factual information is untimely pursuant to section 351.301(b)(2) of the Department's regulations.

Rejection of New Factual Information, PD 395 at bar code 3111617–01 (Dec. 21, 2012). Here, Defendant elaborates on Commerce's reasoning. It explains that VASEP's characterization of the Subsequent Review GNI Lists as information to value factors of production was improper because Commerce does not have to consider relative economic comparability and the economic comparability of the potential surrogate does not factor into Commerce's finding of what constitutes the "best available information" for calculating individual surrogate values.

Commerce categorized the Subsequent Review GNI Lists as "factual information" under 19 C.F.R. § 351.301(b)(2) as opposed to "publicly available information to value factors" under 19 C.F.R. § 351.301(c)(3). It is undoubtedly true that the selection of the primary surrogate country is central to Commerce's selection of sources to value a respondent's factors of production. Thus, it follows that factual information submitted regarding the economic comparability of the primary surrogate country is at least indirectly tied to valuing a respondent's factors of production. However, it is a reasonable application of 19 C.F.R. 351.301(c)(3) that "publicly available information to value factors" refers to information that will be used to directly value a specific factor, rather than indirectly value a specific factor because it influences the choice of the primary surrogate country. Thus, the court cannot say that Commerce arbitrarily or unreasonably applied its regulations. Moreover, in light of Commerce's broad discretion in establishing rules and procedures including enforcing time limits, the court cannot say that Commerce abused its discretion. See e.g. Yantai Timken Co. v. United States, 31 CIT 1741, 521 F. Supp. 2d 1356, (2007); Reiner Brach GmbH & Co. v. United States, 26 CIT 549, 206 F. Supp. 2d 1323 (2002).¹¹

The court cannot sustain Commerce's decision to disregard the 2011 GNI data. Commerce must consider record evidence pertaining to economic comparability and its failure to do so here is contrary to law. The statute requires that Commerce must, to the extent possible, use surrogate data from a market economy country or countries that are at "a level of economic development comparable to that of the non-market economy country" and "significant producers of comparable merchandise." 19 U.S.C. § 1677b(c)(4). Further, Commerce must base its analysis on the best available information. 19 U.S.C. § 1677b(c)(1). Here, Commerce had information on the record that informed the precise question in front of it and it chose to disregard it. This decision was contrary to law.

Commerce justifies its position by erecting a straw man, namely that considering the data could only be used to revise the potential surrogate country list. First, Commerce explains that "[r]evising the list of surrogate countries at a later date would be potentially unfair to the parties and create undue administrative difficulties." *Decision Memo* 5. Commerce's implication is that Plaintiffs' only reason to have Commerce consider the 2011 GNI data is to establish a new list of economically comparable countries. While Plaintiffs may or may not have desired such an outcome, they clearly wanted Commerce to choose a country other than the one it ultimately chose. Commerce confuses two separate determinations: (1) creating a list of potential countries; and (2) selecting a primary surrogate country. This confusion can be illustrated by two consecutive sentences in the *Decision Memo*:

The World Bank GNI/GDP data submitted by VASEP was not available on the record of this review for the Department to use at the time when we made our *surrogate country selection*, as it is dated September 27, 2012, after the Department made its *surrogate country determination*. Thus, the *Surrogate Country List* which the Department released on November 22, 2011 con-

¹¹ VASEP also argues that the Subsequent Review GNI Lists are "practices, methodologies, and standards" which the court should order Commerce to consider on remand. *See* VASEP's Br. 24–25. The court declines this invitation. The Subsequent Review GNI Lists are not a practice, methodology, or standard, but rather a product of Commerce's analysis of information including 2011 GNI data from subsequent reviews. None of the cases cited by VASEP are on point. Additionally, the court will not take judicial notice of properly rejected information.

tained the most up-to-date information accessible from the World Bank regarding countries economically comparable to Vietnam.

Decision Memo 4 (emphasis added). Commerce would not have to revise the list of surrogate countries in order to take the 2011 GNI data into consideration when it made its primary surrogate country selection. The 2011 GNI data was record evidence, accepted by Commerce, that Plaintiffs contend could affect the primary surrogate country selection analysis. Specifically, Plaintiffs argue it could affect the analysis of the relative economic comparability of the countries on the list. The court agrees. As will be discussed below, Commerce's premise that relative economic comparability is irrelevant cannot withstand scrutiny. Moreover, it is possible that Commerce could have considered the data to reassess whether Indonesia was economically comparable.

Second, Commerce claims it will not consider the GNI data because "no party timely challenged the list. . . ." *Decision Memo* 4. Commerce noted "that at no time during the surrogate country comment period did any party, including VASEP, submit any new information or in any way contest the Surrogate Country List." *Decision Memo* 5. Commerce seems to argue that in conducting its analysis it will only consider record evidence submitted up to a point in time. Thus, Commerce contends that after the comment period for having a country remain on the OP List, the parties were foreclosed from submitting evidence and making any arguments related to the selection of a primary surrogate country. Defendant contends that it would be unfair for Commerce to have considered the GNI data without offering other interested parties the chance to comment on the data, and it would be inappropriate for Commerce to consider the data without such comment. Def.'s Resp. 23.

Despite these arguments, Commerce clearly did not complete its analysis on May 23, 2012 because it changed its surrogate country selection after the preliminary results were issued on September 12, 2012.¹² Moreover, Commerce considered data submitted after the preliminary results to make its selection. *See supra* n. 7. Commerce was still under an obligation to consider all the record evidence in conducting its analysis, which was obviously ongoing, even if the parties were foreclosed from offering arguments. Therefore, given

¹² VASEP submitted the information on Nov. 20, 2012. Catfish Farmers of America, argued the Subsequent Review GNI Lists should be rejected on Dec. 4, 2012. VASEP responded on Dec. 11, 2012. Commerce agreed with Catfish Farmers of America, rejected the Subsequent Review GNI Lists, and instructed VASEP to refile without them on December 21, 2012. VASEP resubmitted on December 26, 2012. See supra Section I.

that Commerce accepted the GNI data, was still in the process of selecting a surrogate country, and considered data submitted by other parties submitted on the same day, Commerce must consider the GNI data.

The Court has previously rejected Commerce's attempts to disregard record GNI data that should be considered in Commerce's primary surrogate country analysis. See Dupont Teijin Films v. United States, 37 CIT __, __, 896 F. Supp. 2d 1302 (2013) ("Dupont I") and Dupont Teijin Films v. United States, 37 CIT __, __, 931 F. Supp. 2d 1297 (2013) ("Dupont II") (collectively "the Dupont line"). In the Dupont line, which the court finds on point and persuasive, Commerce placed its OP List on the record with six countries that were economically comparable to China based on 2008 GNI data. Dupont I at 1304. The list included India. Plaintiff-petitioners timely submitted 2009 per capita GNI data and argued that it was the best available information on which Commerce should base its economic comparability decision. Id. at 1305. However, Commerce found India was economically comparable based on 2008 GNI data and chose India as the primary surrogate country. Id. In Dupont I, the court found that Commerce's reliance on 2008 GNI data without a reasonable explanation made its determination unsupported by substantial evidence. Id. at 1307. After remand, Commerce provided a new explanation for disregarding the 2009 GNI data, "that although the 2009 GNI data were placed on the record within the time permitted for submission of factual information, they were submitted too late in the proceedings to be considered by the OP when making its list of economically comparable countries." Dupont II at 1300. The court rejected this rationale explaining that administrative burdens could not excuse Commerce from complying with statutory obligations including determining accurate dumping margins and using data from an economically comparable country. Id. at 1305. The court further explained that Commerce could create a new deadline for parties to submit GNI data, prior to the deadline for the submission of factual information, pursuant to the APA if administrative constraints so required. Id. at 1307.

Here, like in the *Dupont* line, Commerce seeks to justify its refusal to consider 2011 GNI record evidence based on administrative constraints. There is nothing in the administrative record of this case that makes the reasoning in the *Dupont* line less compelling. In fact, here, Commerce's justification, that "the selection of potential surrogate countries [is] similar to the selection of mandatory respondents in an administrative review – both are very important to the proceeding and must be resolved early in the case in order to provide suffi-

cient time for party participation and the necessary analysis," *Decision Memo* 4, is even further undercut by Commerce's reversal of its primary surrogate selection between the preliminary and final results.

Defendant attempts to distinguish the *Dupont* line based upon the timing of the submission in the instant case. Defendant argues that in the *Dupont* line, the information had been submitted prior to the preliminary results whereas here VASEP did not submit the information until much later in the proceeding. Def.'s Resp. 23. Additionally, in the *Dupont* line, Commerce "had the benefit of interested party comments on the data in both case and rebuttal briefs." *Id.* at 24. Defendant goes so far as to state that "VASEP waited until January 11, 2013, when it filed its rebuttal brief, to put interested parties and Commerce on notice that it regarded the 2011 GNI data as relevant." *Id.* (citation omitted).¹³

The Defendant's timing argument is not only a *post hoc* rationale, it mischaracterizes the sequence of events and identifies a distinction that has no relevance in this case. VASEP originally submitted the data on November 20, 2012 and only resubmitted it on December 26, 2012 after Commerce rejected VASEP's submission of the Subsequent Review GNI Lists. Moreover, in this case it would be nonsensical to rely upon the issuance of the preliminary results as a cutoff date for considering information. Here, Commerce changed its surrogate country selection after it issued the preliminary results. Clearly, Commerce did not finalize its primary surrogate country selection by May 23, 2012, the timeframe Commerce asked for comments and data on economic comparability, or by September 12, 2012, the date of publication for the preliminary results. Moreover, this is not a case where the parties advocated for a country that was not on the original OP List. All parties had the opportunity to submit surrogate value data from Indonesia, Bangladesh, and other countries. In fact, if the parties had relied on Commerce's selection of Bangladesh in the preliminary results, they would not have submitted data or argument concerning the country that was ultimately selected.¹⁴

Commerce cannot ignore record data covering 7 out of 12 months of the POR, which may affect its choice of a primary surrogate country.

¹³ At oral argument, Defendant explained that argument based on the 2011 GNI data was made for the first time in VASEP's rebuttal brief in contravention of 19 C.F.R. § 351.309(c)(2). VASEP responded that this argument was made in the rebuttal brief because VASEP agreed with Commerce's selection of Bangladesh in the preliminary results and was making the argument as rebuttal. The court will not rule on this argument because it is both a *post hoc* rationale for Commerce's disregard of the 2011 GNI data and was not raised until oral argument.

¹⁴ Along these same lines, parties always run the risk of submitting information from a country or countries that is not ultimately selected.

In both the underlying proceeding in the *Dupont* line and here, Commerce accepted evidence on the record which it ignored in making its primary surrogate country selection, and that evidence was submitted after the time period Commerce set for surrogate country selection comments. The important difference between the *Dupont* line and this proceeding is that here, Commerce changed its selection, indicating that the primary surrogate country issue had not been resolved, like "mandatory respondents . . . early in the case." Thus, in this case, Commerce has even less justification for its actions than it did in the *Dupont* line.

Finally, Defendant argues that Commerce's determination in the final results to use Indonesia instead of Bangladesh as the primary surrogate country was based on data considerations, a separate inquiry from determining what countries are at a level of economic development comparable to Vietnam. Def.'s Resp. 24. This argument is unpersuasive and even further undermines Commerce's other justifications. The idea that administrative constraints on the one hand restrict Commerce from considering GNI data that covers 7 out of 12 months of the POR, but do not restrict Commerce from considering new data from each potential primary surrogate country, such as the IAS data to value whole live fish submitted on the same day, is nonsensical. Commerce fails to explain why its primary surrogate country analysis for its preliminary determination considers economic comparability, significant production of comparable merchandise, and comparison of available data and its analysis for its final determination only analyzes data considerations. As such, its determination is not in accordance with law.¹⁵

2. Relative Economic Comparability

The parties dispute whether Commerce was required to, and did in fact, compare the relative economic comparability of the countries on its OP List. VASEP argues that the statute requires such a comparison and one was not done here. *See generally* VASEP's Br. 21–22. Defendant argues that a comparison is not required but, that even if it were, Commerce has fulfilled any such obligation in this case because it "thoroughly analyzed' the remaining surrogate country criteria." Def.'s Resp. 25. The court finds that the record in this case requires Commerce to compare relative economic comparability with the data of the potential surrogate countries and here, Commerce has not done so.

¹⁵ As stated above, no party advocates for a country that was not on the original OP List. Thus, the parties and Commerce were not deprived of the opportunity to submit and consider any data from Indonesia or Bangladesh.

As previously discussed, the statute requires Commerce to value the "factors of production in one or more market economy countries that are (A) at a level of economic development comparable to that of the nonmarket economy country, and (B) significant producers of comparable merchandise." 19 U.S.C. § 1677b(c)(4). Commerce has considerable discretion in applying these statutory requirements. Despite its admittedly broad discretion, Commerce cannot lose sight of the overall statutory goal, *i.e.*, to determine accurate dumping margins. *See CS Wind Vietnam Co.*, 38 CIT __, __, 971 F.Supp.2d at 1277 (citing *Rhone Poulenc*, 899 F.2d at 1191; *see also Parkdale Inten.*, 475 F.3d at 1380. Given this constraint on Commerce's discretion, the court must determine whether Commerce's application of its policy in this case was a reasonable approach to determine accurate dumping margins. The court finds that it was not.

Commerce's obligation to determine accurate dumping margins bounds its discretion. Commerce's general policy is to consider all economically comparable countries to be equally comparable, eliminate countries which are not significant producers of comparable merchandise and then choose the country with the best factors data. This policy may often be a reasonable means to obtain accurate dumping margins; it was not here. Here, Commerce was faced with the "unusual situation of having on the record three sources of information issued by governments, which represent official statements of those governments as to the price of whole live fish -i.e., the *Philip*pines FS, Indonesian AS, and DAM Data sources, and one source from an international organization, relevant to [its] analysis - *i.e.*, the FAO FIGIS Data source." Decision Memo 12. Commerce itself noted that "[w]hile [it] typically do[es] not scrutinize official government statistics in such detail, the necessity to respond to the comments raised by interested parties and to select one of the sources compelled [it] to do so in this case." Id. In light of its obligation to determine accurate dumping margins, the statute requires Commerce to consider the relative economic comparability of the economically comparable countries and to weigh these differences against the strengths and weaknesses of the factors data.

As both parties note in their briefs, the question of relative economic comparability was addressed by this Court in Ad Hoc Shrimp Trade Action Comm. v. United States, 36 CIT ____, ___, 882 F. Supp. 2d 1366 (2012) ("Ad Hoc Shrimp I"). In the underlying administrative proceeding of Ad Hoc Shrimp I, Commerce justified its refusal to compare relative GNIs by stating "the statute does not require Commerce to use a surrogate country that is at a level of economic development *most* comparable to the NME country" *Id.* at 1374. The court rejected this justification, explaining that "Commerce's policy of disregarding relative GNI differences among potential surrogates for whom quality data is available and who are significant producers of comparable merchandise is not reasonable, because it arbitrarily discounts the value of economic comparability relative to the remaining eligibility criteria." *Id.* The court further elaborated that "[b]ecause none of Commerce's three surrogate country eligibility criteria is preeminent, it follows that relative strengths and weaknesses among potential surrogates must be weighed by evaluating the extent to which the potential surrogates satisfy each of the three criteria." *Id.* at 1374–1375.¹⁶

Implicit in the Ad Hoc Shrimp I holding is that Commerce's justification ignores relevant sections of the statute and its overall purpose. Although Commerce is not required to choose a primary surrogate country that is the most economically comparable, Commerce is required to use the best available information to value the factors of production. See, e.g., Nation Ford Chemical Co. v. United States, 166 F.3d 1373, 1377 (Fed. Cir. 1999) (explaining the flexibility provided by 19 U.S.C. § 1677b(c) but stating it "mandates that Commerce value the factors of production on the basis of 'the best available information regarding the values of such factors in a market economy country." (citing 19 U.S.C. § 1677b(c)(1)). Commerce's explanation segregates its choice of the best available information to value the factors of production from its selection of the primary surrogate country.

¹⁶ The court notes that during the course of the remand proceedings the government moved the court to expand the scope of the previously ordered remand so that it could consider new information showing that the mandatory respondent had provided false or incomplete information. See Ad Hoc Shrimp Trade Action Comm. v. United States, 37 CIT __, __, 882 F. Supp. 2d 1377 (2013). The court granted the government's motion. See id. at 1381–82. After remand, Commerce concluded that the mandatory respondent had not shown it was eligible for a separate rate. See Ad Hoc Shrimp Trade Action Comm. v. United States, 37 CIT

_____, ____, 925 F. Supp. 2d 1315, 1317 (2013). The court affirmed this finding. See id. Thus, neither the court nor Commerce had the opportunity to address the issue that the court originally remanded, *i.e.*, the importance of the relative economic comparability. However, in another proceeding before the court challenging Commerce's subsequent administrative review, the court did have occasion to explain its prior remand. See Ad Hoc Shrimp Trade Action Comm. v. United States, 38 __, __, 986 F.Supp.2d 1362 (2014) (explaining that in Ad Hoc Shrimp I the court remanded Commerce's selection of a surrogate country over another surrogate country when the data for each country were only slightly different and the country with the slightly worse data was much closer in GNI and a more significant producer of comparable merchandise). This suggests that the court's point in Ad Hoc Shrimp I was not that Commerce is required to assess the relative economic comparability in each case but only when the facts require. This reading also makes sense in light of the statutory framework. See 19 U.S.C. § 1677b(c) (directing Commerce to value the factors of production from prices or costs in a market economy that is at a comparable level of economic development to the extent possible and use the best available information).

However, these analyses are not wholly distinct. The reason Commerce chooses countries that are economically comparable is so that it can value a respondent's factors of production with information from that country. The statute directs Commerce to value the factors of production based on the best available information from "a market economy country or countries," thus leaving it to Commerce's discretion whether to choose one primary surrogate country. However, as discussed above, Commerce has, by regulation, indicated its preference to use information from one primary surrogate country to value factors of production, *see* 19 C.F.R. § 351.408(c)(2), and it did so here. *See Decision Memo* 30 (surrogate financial ratios); *Decision Memo* 31 (labor); *Decision Memo* 32 (sawdust); *Decision Memo* 33 (rice husk). The court expresses no opinion as to whether Commerce would be required under all circumstances to consider relative GNI in relation to data quality differences.

Commerce dismisses the need to address relative economic comparability in this case, relying on its analysis of producers of comparable merchandise and data. Below, Commerce explained that

[w]hile the Department continues to disagree with the Court that the statute requires it to compare relative GNI comparability in its analysis, in these reviews, the Department nevertheless has thoroughly analyzed whether the potential surrogate countries are significant producers of comparable merchandise and the availability of whole fish data in certain of the potential surrogate countries. Therefore, we view the scenario that the Court addressed in *PRC Shrimp Remand* is [sic] distinct from the instant reviews.

Decision Memo 5. Based on the preceding, Commerce continued to treat all countries on the OP List as equally economically comparable. *Id.* at 6. Although Commerce claims that it "thoroughly analyzed" the statutory criteria, the record does not reveal that it did.

In its comparison of the data for whole live fish, the factor with the most influence on respondents' NV, Commerce noted that it "typically [does] not scrutinize official government statistics in such detail" *Decision Memo* 12. Thus from the outset, Commerce indicated the unusual level of scrutiny it would need to apply to distinguish between otherwise usable data sets. Notably, nowhere did Commerce find that the *DAM Data* did not satisfy any one of its criteria (i.e., contemporaneous, publicly available, tax and duty exclusive, representative of a broad market average, and specific to the input). Instead, Commerce found that *DAM Data* did not represent as broad a market average as *Indonesian AS*, was not as specific as *Indonesian*

AS and gave Commerce concerns about the extent to which DAMData was examined for errors. As discussed more fully below, Commerce found *Indonesian* AS and DAM Data to be equally contemporaneous despite the fact that *Indonesian* AS covered the entire POR as well as 12 months outside the POR and DAM Data covered the POR precisely. Here, like in Ad Hoc Shrimp I, the record does not belie a finding that a comparison of the DAM Data and Indonesian ASmakes the choice of Indonesia so clear cut that weighing the relative GNIs of the countries would not improve Commerce's selection of the best available information.

Defendant responds to Plaintiffs' arguments by reiterating Commerce's explanations, and arguing that Commerce's practice of considering all countries on an OP list to be economically comparable "recognizes that the 'level' in an economic development context necessarily implies a range of per capita GNI, not a specific per capita GNI." Def.'s Resp. 27 (citing Policy Bulletin 04.1; *Decision Memo* 3–4). Defendant further argues Commerce found that Indonesia was at the same level of economic development, a finding that satisfies the statutory requirement for using a country at a level of economic development comparable to the NME. Defendant also argues that the statute does not prohibit Commerce from relying upon data quality to choose among otherwise qualified countries. Defendant also characterizes Commerce's analysis as finding Bangladesh data unreliable and Indonesian data superior to Bangladesh data.

These arguments are unpersuasive for several reasons. First, Plaintiffs do not argue that Commerce should blindly choose the primary surrogate country based on which country is the "most" economically comparable. Nor does the court suggest that the analysis should hinge on which country is the "most" economically comparable. As discussed above, the ultimate question is what is the best available information to value the respondents' factors of production? Thus, Commerce must choose the country that furthers this goal. The analysis suggested by *Ad Hoc Shrimp I*, and adopted here, is that Commerce must compare differences in economic comparability with differences in the other factors, including data quality, when the facts so require.

Second, Defendant ignores the statutory framework by isolating Commerce's economic comparability finding. As discussed above, that is not what the statute and regulatory scheme imposed by Commerce, through its preference of valuing factors from a single country, provide.

Third, Defendant's argument that Commerce found Bangladesh data unreliable, mischaracterizes Commerce's actual findings as pro-

vided in the Decision Memo. Commerce did find that DAM Data for whole live fish was less broadly representative, less specific, and less reliable than Indonesian AS data for whole fish, but it did not find that DAM Data for whole live fish was not broadly representative, not specific or unreliable. If Commerce had found DAM Data for whole live fish was unreliable the court would consider whether Commerce's finding in that regard was supported by substantial evidence, but here Commerce did not make any such finding. Additionally as discussed below, Commerce determined to make relative comparisons with respect to some of its criteria but refused to do so for contemporaneity where DAM Data for whole live fish covered the POR precisely and Indonesian AS data for whole live fish covered the POR and an additional 12 months. Finally, Commerce also explained that it considered the data for other factors including, for example, the surrogate financial ratios. In its analysis for choosing DSFI's financial statement as the best available information to value financial ratios. Commerce specifically refused to discuss financial statements from Bangladesh "because [it had] a surrogate financial statement from the primary surrogate country" Decision Memo 30. Thus, it is unclear if Indonesia provided better information to value financial ratios.¹⁷ A similar rationale was provided for several other surrogate values as discussed in Section II.B in this opinion. See infra section ILB.

Defendant also cites to *Jiaxing Bro. Fastener Co. v. United States*, 38 CIT ___, 961 F. Supp. 2d 1323 (2014) to argue Commerce's explanation that it must select its primary surrogate country early in the process is valid. In *Jiaxing Bro.*, the plaintiff challenged Commerce's methodology of using per capita GNI to measure economic comparability as opposed to a method proposed by the plaintiff. *Id.* at 1329–30. The court found Commerce's methodology to be a reasonable reading of the statute under the second prong of *Chevron. Id.* However, as

¹⁷ In the *Decision Memo* Commerce further explains "[c]ombined with the far greater contemporaneity and demonstrated importance to the NV calculation of the additional FOP available from Indonesia we have, consequently, selected Indonesia as the primary surrogate country because it is economically comparable to Vietnam, is a significant producer of comparable merchandise, and has the best available information with which to value the main input into the subject merchandise, whole live fish." *Id.* at 27. This finding does not save Commerce's analysis. As discussed in an above example, Commerce refused to compare data to value surrogate financial ratios, labor, and other factors of production from Indonesia and Bangladesh based on its practice to value all factors in the primary surrogate country. It cannot base its entire comparison on the greater contemporaneity of the Indonesia data without considering its other criteria. Without greater elaboration, the court cannot review the agency's rationale for finding Indonesia to be the primary surrogate country.

explained in *Clearon Corp.*, the challenge at issue in *Jiaxing Bro.*, was to the initial placement of a country on the potential surrogates list, not "the merits of each of the potential surrogates on the list relative to each other." *Clearon Corp.*, Slip Op. 14–88, 2014 WL 3643332, at *10–12. Moreover, the court in *Jiaxing Bro.*, also expressed agreement with the *Dupont* line and *Ad Hoc Shrimp I. Id.* at 1329 (citing *Dupont I* at 1306–1310; *Ad Hoc Shrimp I* at 1374–76)).

The court takes no position as to the relative economic comparability of any of the countries on the OP List as compared to Vietnam. Additionally, the court does not take any position as to how the OP List countries' relative differences in GNI compare to any relative differences in data quality. These findings are for Commerce to make in the first instance. However, on remand, Commerce should consider all GNI data on the record in connection with its other findings relating to the selection of the primary surrogate country, including data considerations. All these findings should be made with the ultimate goal in mind, valuing the respondents' factors of production using the best available information so as to calculate accurate dumping margins.

B. Primary Surrogate Country: Nonfish FOPs

Plaintiffs argue that Commerce's finding regarding the importance of non-fish FOPs was unreasonable. Further, in analyzing the contemporaneity of these non-fish FOPs, Plaintiffs claim Commerce acted contrary to law. Plaintiffs rely upon both of these arguments to challenge Commerce's selection on Indonesia as the primary surrogate country. The court disagrees with Plaintiffs on its first challenge, but finds Plaintiffs have demonstrated that Commerce acted contrary to law in analyzing the contemporaneity of these non-fish FOPs.

1. Importance of Non-Fish for FOPs

Plaintiffs argue that Commerce's reliance on the contemporaneity of nonfish factors in its primary surrogate country selection was not supported by substantial evidence. Specifically, Plaintiffs argue that Commerce's analysis regarding the significance of non-fish FOPs "failed to consider critical information" and thus was unreasonable. Anvifish's & Vinh Quang's Br. 12. Substantial evidence exists on the record when there is "such relevant evidence as a reasonable mind might accept as adequate to support a conclusion." *Nippon Steel Corp. v. United States*, 337 F.3d 1373, 1379 (Fed. Cir. 2003) (quoting Consol. Edison Co. v. NLRB, 305 U.S. 197, 229, 59 S.Ct. 206, 217, 83 L.Ed. 126, 140 (1938)). However, the "substantiality of evidence must take into account whatever in the record fairly detracts from its weight." Universal Camera Corp., 340 U.S. at 488. Nevertheless, "the possibility of drawing two inconsistent conclusions from the evidence does not invalidate Commerce's conclusion as long as it remains supported by substantial evidence on the record." *Zhaoqing New Zhongya Aluminum Co. v. United States*, 36 CIT __, __, 887 F. Supp. 2d 1301, 1305 (2012) (*citing Universal Camera Corp.*, 340 U.S. at 488). Substantial evidence also requires that Commerce "provide a 'rational connection between the facts found and the choice made . . . [and] articulate a satisfactory explanation for its action." *Baroque Timber Industries (Zhongshan) Co. v. United States*, 38 CIT __, __, 971 F. Supp. 2d 1333, 1339–40 (2014) (citing *Burlington Truck Lines, Inc. v. United States*, 371 U.S. 156, 168 (1962); *Yangzhou Bestpak Gifts & Crafts Co. v. United States*, 716 F.3d 1370, 1378 (Fed. Cir. 2013)).

As discussed above, in NME proceedings, Commerce generally calculates NV "on the basis of the values of the factors of production utilized in producing the merchandise" together with "an amount for general expenses and profit plus the cost of containers, coverings and other expenses," 19 U.S.C. § 1677b(c)(1). Commerce seeks "the best available information regarding the values of such factors . . . in one or more market economy countries that are (A) at a level of economic development comparable to that of the nonmarket economy country. and (B) significant producers of comparable merchandise." 19 U.S.C. § 1677b(c)(1)-(4). In making this determination, Commerce prefers to value all factors in a single surrogate country. See 19 C.F.R. § 351.408(c)(2).¹⁸ To select a primary surrogate country, Commerce examines whether a country is economically comparable, a significant producer of comparable merchandise, and whether information on the record from that country is suitable for valuing factors. In comparing data sources from different countries, it may be the case that one country provides better data for one factor and a different country provides better data for another factor. Thus, when selecting which country should be the primary surrogate. Commerce considers which factor(s) contribute the most to the NV calculation.

Here, unlike in past reviews, Commerce determined that nonfish factors were important to the NV analysis stating that "factors other than whole fish and surrogate ratios account for a significant portion of direct materials and NV" *Decision Memo* 10. This conclusion was based on findings that Vinh Hoan was substantially integrated,

¹⁸ In its regulations, Commerce includes factors of production as well as other expense information in its definition of the terms "factors." *See* 19 C.F.R. § 351.408(c) ("Valuation of Factors of Production. For purposes of valuing the factors of production, general expenses, profit, and the cost of containers, coverings, and other expenses (referred to collectively as "factors") under section 773(c)(1) of the Act the following rules will apply").

many Vietnamese exporters of subject merchandise had similar production experiences to Vinh Hoan, and Vinh Hoan was one of the largest exporters of subject merchandise.¹⁹ See id. at 9. Thus, Commerce explained that in addition to its comparison of the record evidence for whole live fish and financial ratios, "the far greater contemporaneity and demonstrated importance to the NV calculation of the additional FOP [sic] available from Indonesia" supported its primary surrogate country selection. *Decision Memo* 27; *see also id.* at 10 (finding that with two exceptions all surrogate values for Indonesia were more contemporaneous than their Bangladeshi counterparts).

Commerce's determination regarding the importance of nonfish factors was in part based upon its analysis of Vinh Hoan, one of the two mandatory respondents, and one of the largest producers. See Decision Memo 9: see also Final Results Analysis Memorandum for Vinh Hoan 2, CD 261 at bar code 3124243-01 (Mar. 13, 2013). Anvifish, the other mandatory respondent, was not an integrated respondent as evidenced by Commerce not using any factors of production from the farming stage in the calculation of Anvifish's NV. Anvifish's & Vinh Quang's Br. 7; see also Preliminary Results Analysis Memorandum for Anvifish 5, CD 204 at bar code 3095040-01 (Aug. 30, 2012). Moreover, Anvifish and Vinh Quang argue, and Defendant fails to refute, that Commerce's statement that the other separate rate respondents were integrated like Vinh Hoan is unsupported. Nonetheless, the court cannot say that Commerce's conclusion regarding the importance of nonfish factors was unsupported by substantial evidence. Although Anvifish is one of the largest exporters, so is Vinh Hoan. Vinh Hoan is integrated, making nonfish factors more significant to Commerce's analysis. The question may be close, but the court cannot say that Commerce's finding is unreasonable.

Respondents below argued that Commerce should continue to "dismiss[] Petitioners' arguments concerning the valuation of" nonfish FOPs "because those inputs are minor." *Decision Memo* 9. Commerce chose not to treat nonfish FOPs as minor because Vinh Hoan, as an integrated producer, produced "an increasingly significant volume of whole fish for use in the production of subject merchandise." *Decision Memo* 9. Implicit in this statement is that even if other producers from Vietnam are not integrated, the fact that one of the largest producers is integrated makes nonfish factors of production more than a minor consideration. While Commerce could have more explic-

¹⁹ Commerce explained that being substantially integrated means "produc[ing] an increasingly significant volume of whole fish for use in the production of subject merchandise." *Decision Memo 9.*

itly acknowledged that its analysis did not depend on both mandatory respondents being integrated, the court finds that Commerce's determination to consider the nonfish FOPs "as a significant portion of direct materials and NV" is supported by substantial evidence. *See Ceramica Regiomontana, S.A. v. United States*, 810 F.2d 1137, 1139 (Fed. Cir. 1987) ("A court may uphold [an agency's] decision of less than ideal clarity if the agency's path may reasonably be discerned.") (citations omitted). Commerce selects a primary surrogate country in order to value the factors of production based on the best available information. It is reasonable for Commerce to consider all the information that will be used to calculate NV under the NME methodology. One of the two mandatory respondents is integrated and therefore nonfish factors will play a role in determining the NV for Vinh Hoan and the separate rate respondents through Commerce's separate rate methodology.

2. Analysis of Nonfish FOP Data

Plaintiffs argue that in analyzing the data for non-fish FOPs Commerce acted contrary to law. Commerce relied in part on the contemporaneity of nonfish factors of production data in making its primary surrogate country selection. As discussed elsewhere in this opinion, Commerce prefers data that is "contemporaneous, publicly available, tax and duty exclusive, representative of a broad market average, and specific to the input." Decision Memo 19 (citation omitted); see also Decision Memo 32 (discussing additional criteria including preferring non-export price averages and non-aberrational data). However, here, in relying on FOPs other than whole live fish and surrogate ratios to make its primary surrogate country selection, Commerce stated that the majority of Indonesian surrogate values were more contemporaneous than Bangladeshi surrogate values. Decision Memo 10. Commerce's failure to analyze any of its other criteria makes its analysis of the data contrary to law.²⁰ Indeed, this Court has on occasion explained that Commerce may not rely on contemporaneity alone in selecting what constitutes the best available information. See, e.g., Hebei Metals & Minerals Import and Export Corp. v. United States, 29 CIT 288, 300-01, 366 F. Supp. 2d 1264, 1274-75 (2005); see also Yantai Oriental Juice Co. v. United States, 26 CIT 605, 616-617 (2002). Thus, on remand, if Commerce continues to rely on factors other than whole fish to make its primary surrogate country selec-

²⁰ Commerce's failure to consider the relevant criteria caused it to disregard detracting evidence that it needs to address to satisfy the court that its determination is supported by substantial evidence.

tion, it must explain why Indonesian data for these nonfish FOPs are the best available information and not only credit their contemporaneity.²¹

Here, Defendant responds that "Commerce examined the record data related to a number of factors of production – financial ratios, labor, sawdust, rice husk, fish waste, fish belly, fish skin, fish oil, fish meal, frozen broken meat, and fresh broken meat - to determine whether they satisfied Commerce's surrogate value criteria." Def.'s Resp. 5253 (citing Decision Memo 27-41). Defendant's response misses the point of Plaintiffs' challenge. Plaintiffs challenge Commerce's failure to meaningfully compare and evaluate the data from all countries based upon all the criteria. Anvifish's & Vinh Quang's Br. 18-19. To say that Commerce satisfied itself that the data met some minimum standard does not respond to the allegation that there has been no meaningful comparison of Indonesian and Bangladeshi data. An examination of the *Decision Memo* reveals that Commerce did not evaluate criteria but often relied upon the fact that Commerce had chosen Indonesia as the primary surrogate country. For example, Commerce selected an Indonesian company's financial statement to calculate financial ratios. Decision Memo 27-30. Commerce based this selection on its finding that the financial statement satisfied its criteria and was from the primary surrogate country.²²

Commerce declined to compare any other evidence on the record. For labor, Commerce selected Indonesian ILO Chapter 5B data. *Decision Memo* 30–31. It explained that although this data was not the most contemporaneous on the record, it satisfied all of the other criteria and was from the primary surrogate country. Commerce declined to compare any other record evidence. For sawdust, Commerce selected Indonesian GTA import data. *Decision Memo* 32. Commerce explained its selection by stating that Bangladeshi UN Comtrade data and price quotes were not as contemporaneous and

²¹ VASEP additionally argues that contemporaneity should not be dispositive because "with respect to labor, energy, transport, by-products and financial ratios, Bangladesh data is either equal to or more contemporaneous than the other data sources." VASEP's Br. 58. Commerce should explain how it found that the majority of Indonesian surrogate values were more contemporaneous than Bangladeshi data in light of VASEP's claim. VASEP goes on to argue that Bangladeshi data for nonfish FOPs and financial ratios is superior to Indonesian data for reasons other than contemporaneity. As Commerce did not analyze factors other than contemporaneity for this data, the court cannot asses this claim.

²² To say that the financial statement, or any other data, "satisfied its criteria" is to adopt a "check the box" approach. In other words, Commerce would be able to prefer data from a particular country without any comparison amongst countries so long as the data satisfied some minimum requirement. Given Commerce's reliance on the data to select the country in the first place, this approach is not consistent with Commerce's obligation to use the best available information to calculate accurate dumping margins.

there was information from the primary surrogate that satisfied its criteria. For rice husk, Commerce selected Indonesia GTA import data. Decision Memo 33-34. It explained that this data was from the primary surrogate country and satisfied its criteria. Additionally, Bangladeshi price quotes were not the best available information, "because, among other reasons, they are not contemporaneous" Decision Memo 34. For fish waste, fish belly, and fish skin, Commerce selected a price quote from a Philippine company. Decision Memo 35. For fish oil, Commerce selected Indonesian GTA import data, but determined to cap the value because of concerns that the tariff heading was overly broad. Decision Memo 37-39. It did not discuss this data in comparison to data from any other surrogate country, but did reject Indonesian price quotes as unreliable. For fish meal, Commerce selected Bangladeshi UN Comtrade data over an Indonesian price quote because it found the Indonesian price quote to be unreliable. Decision Memo 38-39. For frozen broken meat, Commerce selected Indonesian GTA data but did not compare the data to data from any other potential surrogate country. Decision Memo 39-40. For fresh broken meat, Commerce selected a price quote from a Philippine company over Indonesian GTA data because the price quote was more specific. Decision Memo 40-41.

As demonstrated by this summary, Commerce selected information from either the Philippines or Bangladesh for five of the eleven surrogate values cited by Defendant: fish waste, fish belly, fish skin, fish meal and fresh broken meat. Additionally, for two of the six Indonesian based surrogate values, financial ratios and labor, Commerce did not compare the Indonesian data with data from other countries because it had data "from the primary surrogate country." Decision Memo 30, see also id. at 31. For sawdust and rice husk Commerce similarly relied on its practice of using data from the primary surrogate country, but also added that Indonesian data was more contemporaneous than Bangladeshi data. For the final two surrogate values cited by defendant, frozen broken meat and fish oil, Commerce did not discuss data from any other country, presumably because no party argued for data from those countries below. The only surrogate value for nonfish factors cited by Defendant that could potentially be construed to include a discussion of criteria other than contemporaneity is rice husk where Commerce chose Indonesian data over Bangladeshi data "because, among other reasons, they are not contemporaneous" Decision Memo 34 (emphasis added). Commerce's use of the phrase "among other reasons" and its discussion of contemporaneity for one other factor cannot reasonably be considered to be comparison of Indonesian data with data from other potential surrogate countries. Moreover, Commerce cannot both rely on nonfish factors to make its primary surrogate country and select the surrogate value for each nonfish factor based on its primary surrogate selection. That would be circular reasoning. This is not enough to sustain Commerce's selection of Indonesia.

C. Primary Surrogate Country: Financial Ratios

As a separate matter, Anvifish and Vinh Quang challenge Commerce's selection of Indonesia as the primary surrogate country because Commerce's analysis of financial ratios is both contrary to law and unsupported by substantial evidence. The court finds that Commerce's reliance on circular reasoning in selecting financial ratios undermines its primary surrogate country selection. As for the Plaintiffs' arguments that the financial ratio analysis is not supported by substantial evidence, because Commerce relies on its primary surrogate country selection to choose the Indonesian financial statement as the best available information, the court must reserve consideration of this issue until Commerce's decision after remand. Commerce must compare financial statements in the first instance.

Anvifish and Vinh Quang argue that Commerce's failure to compare financial statements from Indonesian and Bangladeshi companies made its surrogate country selection contrary to law. Anvifish's & Vinh Quang's Br. 19. Defendant responds that Commerce did compare financial statements because it "regarded contemporaneity of Indonesian data for 'all [surrogate values]' an 'important factor' in its analysis." Def.'s Resp. 53 (citation omitted). Such a broad conclusory statement regarding surrogate values in general cannot be construed as a comparison. Further, Defendant claims that "Anvifish has failed to demonstrate that it was unreasonable for Commerce to follow its regulatory preference for valuing all factors of production using reliable data from a single country." Def.'s Resp. 53 (citation omitted). By stating that it relies upon a comparison of financial statements in making its primary surrogate country selection and explaining that Indonesia provides the best available information to calculate financial ratios because it is from the primary surrogate country, Commerce has placed respondents on an administrative pendulum swinging back and forth between two explanations that rely on each other.

Commerce did not consider financial statements from Bangladeshi or Philippine companies "because [it had] a surrogate financial statement from the primary surrogate country which [met] the Department's SV selection criteria" *Decision Memo* 30. Commerce cannot base its primary surrogate country decision in part on financial statements but base its selection of a financial statement solely on its primary surrogate country selection.

D. Primary Surrogate Country Selection: Whole Fish

VASEP argues Commerce's conclusion that IAS data was superior to DAM Data for valuing the whole live fish input is unsupported by substantial evidence and contrary to law. Moreover, Commerce's unsupported findings and legal errors regarding the IAS and DAM data undermined Commerce's primary surrogate country selection because Commerce's whole fish analysis was a major factor in Commerce's ultimate selection of Indonesia. VASEP's arguments can be generally grouped into issues regarding whether the data sets are: (i) specific to the whole live fish input (*i.e.*, similarity of the merchandise and industry covered by the data sets), (ii) contemporaneous, (iii) a broad market average, and (iv) reliable. As an initial matter, given Commerce's obligation to evaluate data in light of the relative economic comparability of the potential surrogate countries, the court cannot say whether Commerce's determination regarding the selection of Indonesia based on the superiority of Indonesian data is reasonable or in accordance with law. Commerce must first address the court's concerns described above including whether any differences in the GNIs and their interaction with the data considerations will affect Commerce's selection. Nonetheless, even if Commerce were permitted to examine data without reference to relative economic comparability, there are several problems that call into question the reasonableness and legal sufficiency of Commerce's analysis. Therefore, upon remand, Commerce should consider the issues discussed below when evaluating and comparing the data and relative economic comparability.

As discussed above, Commerce's policy where there is more than one country that is economically comparable and a significant producer of merchandise is to select "the country with the best factors data." *See* Policy Bulletin 04.1. Here, Commerce compared record information from the Philippines, Indonesia, and Bangladesh for the whole live fish FOP, eventually concluding that IAS data was the best available information.²³

²³ Commerce also considered data for financial ratios and nonfish FOPs, but as discussed supra, the court cannot consider the reasonableness of Commerce's analysis of financial statements because Commerce did not attempt to actually compare financial statements. Similarly, for nonfish factors, Commerce did not conduct an analysis of relevant criteria but merely declared the contemporaneity of the Indonesian data to be superior. The court cannot evaluate Commerce's analysis for reasonableness until it employs a methodology which accounts for the criteria relevant to the analysis.

The court also reserves judgment on Plaintiffs' arguments that the surrogate values chosen for financial ratios, labor, sawdust, rice husk, fish waste, fish belly, fish skin, frozen

VASEP argues that IAS data is less specific than DAM data because IAS data includes non-subject merchandise species of pangasius (*i.e.*, jambal) and the Vietnamese industry is more comparable to Bangladesh's industry than to Indonesia's industry. See VASEP's Br. 30-36, 49-50. VASEP argues that including jambal distorts the price for whole live fish in IAS data because jambal is higher priced than species which are subject to the dumping order, such as hypophthalmus. Further, IAS data is distorted by including alternative aquaculture methods (i.e., cages, rice paddies, and floating nets) that are distinct from those used in Vietnam (*i.e.*, ponds) because there are different cost structures for all of these methods. VASEP also argues the Indonesian industry is distinct from Vietnam and Bangladesh because unlike farmers in those countries, farmers in Indonesia further process whole live fish before shipping the fish. Also, Indonesia imports subject merchandise and exports only small amounts of subject merchandise whereas both Vietnam and Bangladesh export substantial amounts of subject merchandise. VASEP also argues that Commerce's concerns that DAM data was less specific than IAS data because DAM data included dead fish was not supported by substantial evidence. Finally, VASEP argues that prices from Indonesia are distorted by subsidies, which again makes whole live fish grown in Indonesia less specific than whole live fish in Bangladesh.

Commerce addressed these arguments below. Regarding VASEP's argument that the IAS data is distorted by inclusion of higher priced jambal, Commerce stated:

[i]n this review, IAS officials have stated that there [sic] only two types of pangasius commonly grown in Indonesia, hypophthalmus and jambal, with hypophthalmus being the predominant species grown. In addition, IAS officials have stated that hypophthalmus grows in ponds and cages, while jambal grows in rivers. Therefore, by limiting the aquaculture area to ponds and cages, only pangasius hypophthalmus, and trace amounts of the pangasius hypophthalmus hybrid, are represented in the Indonesian AS data. There is no evidence that the inclusion of the pangasius hypophthalmus hybrid distorts prices significantly. In fact, because hypophthalmus and jambal sell at similar

broken meat, fresh broken meat and freight in were unsupported by substantial evidence. For each of these surrogate values, Commerce either explicitly states that it is basing its choice, in part, on its primary surrogate country selection, or its choice may be affected by a change in its choice of primary surrogate country. Therefore, the court must reserve its substantial evidence analysis for after Commerce has had the chance to reconsider its primary surrogate country selection consistent with this opinion.

Here, with respect to whole fish, and whether Commerce's preference for Indonesian data supports its selection of Indonesia as the primary surrogate country, Commerce made a comparison between the various potential sources on the record. Thus, the court is able to evaluate whether Commerce's analysis was reasonable.

prices, it is reasonable to suspect that the hybrid would sell at similar levels and that the inclusion of any such prices in the data would not distort the reported prices significantly.

Decision Memo 24 (footnotes omitted). Commerce cited a statement from Director General of Aquaculture Soebjakto ("Soebjakto Statement"), which was attached to an affidavit submitted by petitioners, CFA Surrogate Value Data Submission Ex. 6 at Att. 5, PD 335 at bar code 3106818-21 (Nov. 20, 2012) ("Sopaheluwakan Aff."), to support its first two findings. To answer the question posed by Catfish Farmers of America, "[i]s there any one pangasius species predominantly grown in Indonesia? (As we understand by far, Pangasius Hypophthalmus is the predominant species produced in Indonesia. Is it true?)," Director General Soebjakto responded "[v]es, it is true. Since 1980s, the pangasius species predominantly grown is [hypophthalmusl widely raised in freshwater ponds and cages culture. However, [] jambal which has better quality of flesh (texture and colour) is developed since 2000s and grown in streams." In making the finding that hypophthalmus and jambal sell at similar prices, Commerce cited the IAS Data, which Defendant explains shows similar prices for ponds and cages (presumed to be hypophthalmus) versus floating net (presumed to be jambal) in the IAS data.

To support its arguments, VASEP cites an affidavit from the Director General of Aquaculture during the POR, Ketut Sugama, indicating that the IAS data can include several pangasius species, including both hypophthalmus and jambal. *See* VASEP's First Surrogate Value Rebuttal Submission Ex. 19E, PD 212 at bar code 3081393–03 (June 14, 2012) ("Sugama Aff."). It also cites three articles relating to the cultivation of jambal in Indonesia, VASEP's First Surrogate Value Rebuttal Submission Exs. 19B, 19C, 19I, PD 212 at bar code 3081393–03 (June 14, 2012), and the Sopaheluwakan Affidavit as well as two other documents, to show that jambal is higher priced than hypophthalmus, VASEP's Post-Preliminary Surrogate Value Rebuttal Submission Exs. 6, 7A, PD 367–68 at bar code 3108726–03–04 (Dec. 4, 2012). Moreover, VASEP argues that Commerce's citation to the IAS data does not support its finding that hypophthalmus and jambal sell at similar prices.

Commerce addressed two of the articles VASEP argues support the assertion that jambal is grown in ponds. Commerce stated

[r]egarding the argument that jambal is grown in ponds, we disagree. VASEP has cited two articles in support of this contention. One is a guide on how a fish farmer would go about qualifying for a program to grow jambal, and notes that jambal could be grown in ponds. We note that this article is dated to 2006 and does not indicate that jambal was grown in ponds during the POR. The second article cited by VASEP is a technical guide on how jambal could be grown in ponds and cages. We note that this article is dated to 2005 and does not indicate that jambal was grown in ponds during the POR. Due to the age of these articles, we find the IAS official's statement that jambal is grown in rivers to be more probative than the articles cited by VASEP in determining the aquaculture in which jambal is current [sic] grown in Indonesia.

Decision Memo 24–25 (footnotes omitted). Moreover, Defendant in its brief to the court points out that the Soebjakto statement does not provide that jambal is higher priced than hypophthalmus, but rather provides that jambal "has better quality of flesh (texture and colour). . . ." Soebjakto Statement \P 2.

The court's role is not to reweigh the evidence. Nonetheless, it is unclear to the court how the Soebjakto statement that jambal "is developed since 2000s and grown in streams" means that jambal is grown exclusively in rivers using floating nets (as opposed to cages for example), or that jambal is not grown at all in ponds. Commerce, relying upon the Soebjakto statement, assumes that jambal is grown exclusively in rivers and that cages are not used in rivers. It then infers that jambal is only cultivated in floating nets while hypophthalmus is only cultivated in ponds and cages. Commerce then compares prices of products cultivated in floating nets to prices of products from cages and ponds and concludes that hypophthalmus and jambal are similarly priced. It is difficult for the court to follow, without more explanation or evidence, how Commerce arrives at the assumptions, inferences, and conclusions that it does. It is particularly difficult given that in the same statement upon which Commerce relies we are told that jambal "has better quality of flesh (texture and colour). . . ." Soebjakto Statement ¶ 2. This statement suggests that the products would not be similarly priced, and therefore detracts from Commerce's ultimate conclusion based upon the assumptions and inferences. Commerce should address these issues upon remand.

VASEP cites additional evidence to support its argument that IAS data is distorted (and therefore not specific) because Indonesia is a net importer of subject merchandise, much of which comes from Vietnam. *See* VASEP's Post-Preliminary Surrogate Value Rebuttal Submission Exs. 4–5, PD 367 at bar code 3108726–03 (Dec. 4, 2012).

Along the same lines, VASEP also cites evidence showing that Indonesian farmers process fish before shipping it to processing facilities. see Sugama Aff. ¶ 8, which VASEP argues further distinguishes the industry in Indonesia from Vietnam. VASEP argues Commerce failed to address these arguments. Defendant argues in response that its obligation is to compare merchandise, not industries. Def.'s Resp. 41 (citing Decision Memo 24). Although Defendant is correct that the statute directs Commerce to seek surrogate values from producers of comparable merchandise, its explanation is unresponsive to the argument that Plaintiffs are making. There are several producers of comparable merchandise. See Decision Memo 10 (finding Bangladesh, India, Indonesia, Nicaragua, Pakistan and the Philippines to all be significant producers of comparable merchandise). However, when Commerce compares the data from these producers, it is being asked to weigh the specificity of the competing data so as to determine the best available information on the record to value the input. It is not clear to the court how Commerce has addressed VASEP's argument that data derived from countries that are producers of comparable merchandise, where the merchandise is produced in a manner most similar to the subject merchandise, is more specific. Commerce should address this argument on remand.

Finally, VASEP argues that Commerce's concerns regarding the inclusion of dead fish in the DAM data are unfounded. VASEP argues that the evidence cited by Commerce to show dead fish are included in the DAM data, CFA's Submission of Rebuttal Factor Value Data Ex. 30, PD 194 at bar code 3081368–15 (June 14, 2012), does not in fact discuss the inclusion of live and dead fish in the DAM data. Thus, all Commerce is left with is other evidence that there is dead fish sold in Bangladesh. VASEP argues that Commerce's concerns are unreasonable in light of the evidence, an affidavit from a DAM official stating that DAM data does not include dead fish, VASEP's Surrogate Country Comments and First Surrogate Value Submission 13B, PD 143 at bar code 3076936–04 (May 23, 2012), and general information about the sale of dead fish in Bangladesh.

In addressing these arguments Commerce stated

multiple sources on the record indicate that dead fish may be included in the DAM Data. An affidavit detailing interviews of pangasius traders at two large markets, for which DAM reported data during the POR, notes that live pangasius transported from farms to the marketplace die during transit (in some cases the mortality rate is 50 percent), vendors sell live and dead fish at the markets side-by-side, and dead fish are sold at lower prices than live fish. An article published by the U.S. Agency for International Development indicates that up to 29 percent of the pangasius sold in Bangladeshi wholesale markets are dead, and that dead fish sell for less than live fish. Information placed on the record by VASEP indicates (1) that while precautions are taken to ensure that live fish are delivered to the processing plants, there is a two to three percent mortality rate for those fish; and (2) that live and dead fish are sold in wholesale markets. Moreover, there are two competing affidavits on the record from DAM officials concerning whether dead fish are included in the DAM Data. One affidavit, submitted by Petitioners, indicates that dead fish have been included in the DAM Data. Another affidavit, submitted by VASEP and which concerns the DAM Worksheets, indicates that only live fish are included in the DAM Data. It is precisely because of this conflicting information that the Department requested that DAM clarify this issue. However, although provided two opportunities in this administrative review, DAM did not respond to the Department's questions, nor did DAM respond to the Department's questions in the last review. In this case, because DAM has not responded to the Department's questions concerning whether dead fish are included in the DAM Data, we cannot discern with certainty whether and to what extent the DAM Data represents prices only for whole live fish. The respondents have all indicated that they only consume live whole fish. Consequently, we do not find the DAM Data to be as specific as another source on the record. Moreover, because dead fish sell for less than live fish, we harbor concerns that the DAM Data may understate the price of the whole live fish FOP. As noted below, the Department does not have this concern with respect to Indonesian AS.

Decision Memo 22-23.

Based upon the record evidence and the explanation provided by Commerce, the court cannot say that Commerce's conclusion on this issue is unreasonable. First, the court notes that while VASEP correctly claims that the evidence cited by Commerce to show that dead fish are included in the DAM data does not in fact discuss dead fish, Commerce appears to have inadvertently cited to the wrong exhibit. Another exhibit, CFA's Submission of Rebuttal Factor Value Data Ex. 30, PD 202 at bar code 3081368-25 (June 14, 2012), appears to be what Commerce meant to cite, and it does discuss dead fish. However, this typographical error aside, Commerce had evidence that led it to believe dead fish may have been included in the DAM data. This possibility raised concerns for Commerce regarding the usefulness of this data. Commerce sought further information from DAM to clarify, and DAM did not respond. As a result, Commerce was not unreasonable in its conclusion that it could not "discern with certainty whether and to what extent the DAM Data represents prices only for whole live fish." *Decision Memo* 23.

VASEP also argues that IAS data lacks specificity because the inclusion of prices from different levels of trade and for different sizes of fish distorts it. VASEP's Br. 38–39, 41–42. Commerce addressed these arguments below stating,

[w]hile VASEP and Vinh Hoan argue that *Indonesian AS* does not contain size data, we note that none of the respondents are able to report their CONNUMs on a size specific basis, nor do they report purchases of whole live fish on a size-specific basis, thus, we do not find a lack of sizing information in *Indonesian AS* to be any less specific than data which does contain size data.

As we noted in the last review, it is uncertain the extent to which [using wholesale prices and not farmgate prices] is relevant in the surrogate valuation analysis. Surrogate valuation seeks to determine the price a respondent would pay for an input if it were to produce subject merchandise in the surrogate country, not necessarily what producers/sellers of the input in the surrogate country receive. Therefore, whether the *FAO FIGIS Data*, *DAM Data*, *Philippines FS* or *Indonesian AS* represents wholesale prices or farmgate prices is immaterial to our SV analysis.

Decision Memo 24-25.

With respect to the different levels of trade, VASEP cites the Sugama Affidavit, which provides that "[t]he prices for [pangasius] reflected in the IAS surveys are from the retail market channel and include markups by intermediary buyers/distributors from the farmgate prices." Sugama Aff. ¶ 6. VASEP acknowledges that the Soebjakto statement attached to the Sopaheluwakan Affidavit provides that IAS data is derived from "multiplication between the number of production and the average pangasius price at the farm gate level," but VASEP argues that the Sugama Affidavit is more probative because he was in charge of collecting the IAS data for the relevant POR, and Soebjakto was only in charge afterwards. VASEP argues that at the very least, it is unclear whether the IAS data is derived from farmgate, wholesale, or a blend of prices, whereas DAM data prices are categorized under grower, wholesaler, and retail market segments. This distinction, according to VASEP, weighs in favor of selecting DAM data.

VASEP does not explicitly argue that the ability to distinguish levels of trade in the data is important for identifying accurate surrogate values. It simply argues that the DAM data is better than the IAS data because it gives Commerce the ability to choose prices from different levels of trade. Commerce explains that it does not need to distinguish between the different levels of trade and VASEP has not refuted this contention. Therefore, the court cannot say that Commerce's determination on this issue is unreasonable.

With respect to the sizes of fish, VASEP argues that Commerce defines CONNUMS by weight (band)-size of the fish and that all factors of production are based on the size of the end product. However, as Commerce explained, respondents do not "report purchases of whole live fish on a size-specific basis" *Decision Memo* 24. Since respondents do not report the input by size, it is unclear why data that includes the price of the input broken down by size is more specific in any way that is relevant to valuing whole live fish. Thus, Commerce's finding is not unreasonable.

Finally, VASEP argues that the IAS prices are less specific because various subsidies the Indonesian government provides to the pangasius industry distorts them. VASEP cites VASEP's First Surrogate Value Rebuttal Submission Ex. 19G, PD 212 at bar code 3081393–03 (June 14, 2012), which provides a list of subsidies provided to the fishing industry in Indonesia. VASEP also cites information which it argues shows the Indonesian Ministry of Marine and Fisheries Affairs provides various assistance to small farmers, VASEP's Post-Prelim Surrogate Value Rebuttal Submissions Ex. 2, PD 365 at bar code 3108726–01 (Dec. 4 2012), and information it argues shows various loan or credit programs, VASEP's First Surrogate Value Rebuttal Submission Ex. 19A, PD 212 at bar code 3081393–03 (June 14, 2012).

Commerce addressed the issue of whether subsidies affected the IAS data below stating,

it is the Department's practice to exclude data from consideration only when the record evidence demonstrates that the alleged subsidy programs constituted subsidies found countervailable by the U.S. government in a trade remedy proceeding. In this case, as we have found in prior segments, there is no record evidence that the alleged subsidies constitute countervailable subsidies. Decision Memo 20 (footnotes omitted). VASEP argues that Commerce's explanation was contrary to Congressional intent, its own policy, and was thus contrary to law. VASEP cites the Omnibus Trade and Competitiveness Act of 1988, H.R. Rep. 100 576, at 59 (1988), reprinted in 1988 U.S.C.C.A.N. 1547, 1623-24; Folding Metal Tables and Chairs from China, 67 Fed. Reg. 20,090, 20,093 (April 24, 2002) (notice of final determination of LTFV); Issues and Decision Memorandum for Tapered Roller Bearings and Parts Thereof from the People's Republic of China Cmt. 1, A-570-601, (Nov. 15, 2001), available at, http://enforcement.trade.gov/frn/summary/prc/ 01-28651-1.txt (last visited February 12, 2015). While Commerce may have previously found distortion by virtue of subsidies in the absence of an affirmative countervailing duty finding in a U.S. trade remedy proceeding, it did so under very different circumstances. Commerce was not unreasonable in finding that the record here did not support a finding in this case that Indonesian data was distorted by subsidies.

VASEP challenges the contemporaneity of IAS data as compared to DAM data and argues that Commerce erred by not giving any weight to the fact that DAM data precisely coincided with the POR whereas IAS data consisted of data 7 months prior to the POR, the entire POR, and 5 months after the POR.²⁴ Rejecting this argument Commerce explained, "we consistently have recognized that data sources which overlap part of the POR are considered contemporaneous." *Decision Memo* 19 (footnote omitted). Defendant further argues that "the statute does not require Commerce to use the 'most-contemporaneous' data in deciding which data represents the 'best available information." Def.'s Resp. 34 (citing 19 U.S.C. § 1677b(c)(1)). Defendant further argues that this practice has been judicially affirmed. *Id.* (citing *Shieldalloy Metallurgical Corp. v. United States*, 20 CIT 1362, 1371 947 F. Supp. 2d 525, 534 (1996)).

The court agrees with the Defendant that the statute does not require Commerce to use the most contemporaneous data. However, finding that data is contemporaneous in some regard is different from saying that the two periods at issue here are equally contemporaneous. On remand, Commerce should weigh the DAM data's greater contemporaneity when comparing the two data sets and determining which set is the best available information.

²⁴ The reason for this is that DAM data provides monthly prices and thus Commerce can use the months that match up exactly with the POR. In contrast, the IAS data represents yearly averages for 2010 and 2011 thus, each year inherently includes data outside the POR (*i.e.*, 2010 includes data for 7 months prior to the POR, January through July, and 2011 includes data for 5 months after the POR, August through December).

Next, VASEP argues that Commerce's concerns regarding whether DAM data represented a broad market average and its finding that IAS is a broader market average than DAM data are unfounded. VASEP states that in comparison to the weekly price points of two sizes of pangasius in the DAM data, IAS data provides estimated annual figures calculated from sample surveys of a small group of selected respondents. VASEP further argues that the availability of weekly price points in DAM data makes the data qualitatively better and "prone to more sophisticated analysis and review." VASEP's Br. 51. VASEP also argues that the absence of the Mymensingh region from the online DAM database does not detract from the broadness of the DAM data because the record contains hard copy DAM data for Mymensingh. VASEP argues that Commerce should not totally discount the hard copy DAM data as not publicly available because Commerce did not point out any distortion caused by the hard copies not yet being published in the online database. Moreover, VASEP states that the DAM data still accounts for 27 districts in the country even without the hard copy data. Finally, VASEP argues that Commerce's determination was contrary to law because it was inconsistent with a recently completed new shipper review covering the same subject merchandise.

Commerce did not find that the DAM data does not represent a broad market average. *Decision Memo* 21. On the contrary, Commerce found that IAS data represents a broader market average than DAM data, comparatively speaking, based upon a comparison of the district and data coverage for the IAS and DAM data. Its finding, in light of the record evidence, is reasonable. ²⁵

Finally, VASEP argues that the record reveals the IAS data was unreliable and that Commerce's concerns with regard to the reliability of DAM data are unfounded. VASEP argues that IAS data is unreliable because it is not based on actual sales prices and contains mistakes and inaccuracies. Moreover, it claims Commerce's suggestion that DAM data may be unreliable because of a difference between the weekly and yearly averages is unsupported. Commerce addressed these arguments below explaining that with respect to price fluctuations in the data,

the DAM Data and Indonesian AS both have price fluctuations. With the exception of one area representing an insignificant quantity, the highest price in the DAM Data is roughly twice as

²⁵ The court notes that Commerce was able to compare the data from Indonesia and Bangladesh with respect to its broad market average criteria. This is precisely the type of comparison that Commerce did not do with respect to contemporaneity of IAS and DAM data and which it should do on remand.

high as the lowest price. However, this is to be expected in different markets with different supply, demand, and logistical characteristics. VASEP cites to a shrimp from the PRC review as evidence that the Department conducts standard deviation tests when comparing SVs. In that case we noted in the SVs memorandum that an interested party conducted a standard deviation and we found that the standard deviation contributed to our finding that a certain shrimp feed SV was aberrational. However, we did not rely upon the standard deviation in the PRC Shrimp Final. As a result, we do not find that any of the live whole fish SV choices discussed above to be anomalous with regard to price variances and, thus, consider all sources equal in this regard.

Decision Memo 26 (footnotes omitted). Moreover, with respect to DAM data, Commerce

found that it contains price differences between the weekly and yearly averages. For example, the POR yearly price for *pangasius* (small) from Tangail is 7,220 Bangladeshi taka, while an average of the POR weekly prices is 7,019 Bangladeshi taka. It is because the Department is uncertain as to what, if any, procedures are used by DAM to ensure its data accuracy, that the Department requested information from DAM concerning its collection and collation methods; however, DAM did not respond.

Id. (footnotes omitted). In contrast, with respect to IAS data, Commerce did "not have the same concerns" Id. Commerce explained that

[d]ata of *pangasius* production is collected in stages at the household, village, and municipal level, using random sampling to determine the surveyed villages and households which conduct *pangasius* aquaculture activities. Moreover, IAS officials indicate that they do make revisions and corrections to data when necessary. For example, IAS officials acknowledged an error with regard to the 2010 paddy aquaculture area, an area which the Department has not used to calculate the whole live fish SV.

Id. at 26–27 (footnotes omitted).

First, VASEP argues that Commerce violated a preference for "actual price data alone," which VASEP says is akin to DAM data, as opposed to "extrapolated estimated averages" which VASEP says is akin to IAS data. VASEP's Br. 36. It cites Commerce's rejection, in the 5th administrative review, of a potential surrogate value for whole live fish derived from total value and production data in a thesis. *Id.* Defendant responds that this argument is non-responsive to Commerce's reliability analysis below and ignores the fact that each segment of a proceeding is sui generis. Def.'s Resp. 48–49. Moreover, Defendant argues that the Federal Circuit has affirmed Commerce's use of data based on averages. *Id.* at 49 (citing *Ad Hoc Shrimp Trade Action Comm. v. United States*, 618 F.3d 1316, 1322 (Fed. Cir. 2010)).

Next, VASEP argues that IAS data contains mistakes and inaccuracies, a point VASEP says IAS acknowledges. To support this contention, VASEP cites an email exchange between counsel and IAS staff members, VASEP's Post-Prelim Surrogate Value Submission Ex. 1, PD365 at bar code 3108726–01 (Dec. 4, 2012), which acknowledges errors in the IAS data. Additionally, VASEP argues that IAS data contains statistical anomalies which it argues are demonstrated by several exhibits which it attached in its post-preliminary rebuttal case brief. *See* VASEP Post-Preliminary Rebuttal Case Brief Exs. 2(a)-(b), 3(a)-(d), 4(a)-(b), PD 416, at bar code 3114445–04 (Jan. 11, 2013).

Finally, VASEP argues that Commerce's concerns with respect to the reliability of DAM data are unsupported and that any concerns should be allayed by the extensive corroborating evidence on the record. VASEP's Br. 54-56. VASEP argues that Commerce's findings that "a nearly doubling of prices between districts" presents no reliability problem but that "a mere 3% deviation in prices" between weekly and yearly averages does present a reliability problem, are contradictory. Id. at 54. Moreover, VASEP argues that Commerce's concerns that the DAM data is not carefully vetted is contradicted by the relative stability of the prices in hundreds of data points throughout the POR. VASEP also points to various evidence which it argues supports that the DAM data is, in fact, carefully vetted. See VASEP's Surrogate Country Comments and First Surrogate Value Submissions Ex. 13A-13C, PD 143 at bar code 3076936-04. Finally, VASEP cites record evidence which it argues corroborates the DAM prices, VASEP's Surrogate Country Comments and First Surrogate Value Submission Ex. 15A at p. 22, 15B, 16A at p. 25, 16B, 17A, PD 143 at bar code 3076936-04 (May 23, 2012) and explains that there is no such corroborating evidence for the IAS prices.

VASEP's arguments are unpersuasive. With respect to both VASEP's arguments regarding the use of actual price data and the lack of a standard deviation analysis, VASEP has not established that either is required to ensure reliability. Commerce has broad discretion in assessing the reliability of the data.²⁶ Further, Commerce's finding that it had unanswered questions about the DAM data is supported by the fact that DAM did not respond to its queries.

On remand, when Commerce undertakes its comparison of Bangladesh and Indonesia with respect to differences in economic comparability and differences in data considerations, Commerce should address the court's above-stated concerns with Commerce's whole live fish analysis.

III. Surrogate Values

A. Legal Framework for Valuing Factors of Production

Again, in NME cases, Commerce constructs NV by valuing factors of production that are used to produce the subject merchandise and adding "an amount for general expenses and profit plus the cost of containers, coverings, and other expenses." 19 U.S.C. § 1677b(c)(1). Commerce must value the factors of production "based on the best available information regarding the values of such factors in a market economy country or countries." Id. When calculating NV, Commerce "may offset production costs incurred by a respondent with the sale of by-products generated during the production process." Decision Memo 34 (citing 19 U.S.C. § 1677b(c); Guangdong Chems. Imp. & Exp. Corp., 30 CIT 1412, 1422, 460 F.Supp.2d 1365, 1373 (2006)); see also Tianjin Magnesium Intern. Co., v. United States, 34 CIT __, __, 722 F.Supp.2d 1322, 1336 (2010) (explaining that "[t]he antidumping statute does not prescribe a method for calculating byproduct offsets instead leaving the decision to the technical expertise of the Department."). Commerce has broad discretion in deciding what constitutes the best available information because the term is not defined in the statute. See QVD Food Co. v. United States, 658 F.3d 1318, 1323 (Fed. Cir. 2011). However, Commerce must ground its selection of the best available information in the overall purpose of the AD statute, calculating accurate dumping margins. See CS Wind Vietnam Co., Ltd. v. United States, 38 CIT , , 971 F.Supp.2d at 1277 (citing Rhone Poulenc, Inc., 899 F.2d at 1191; see also Parkdale Inten., 475 F.3d at 1380.

²⁶ The court notes that there does appear to be a significant anomaly with respect to Sumatra Region Lampung locality in 2010 cage culture. *See* IAS Data 2010 at pp. 43, 46; VASEP Post-Preliminary Rebuttal Case Brief Exs. 2(a), PD 416 at bar code 3114445–04 (Jan. 11, 2013). The court's standard of review precludes it from reweighing the evidence, but it is possible that Commerce may decide on remand that excluding cage data would improve the accuracy of the respondent's margins in light of the court's discussion of jambal above. Alternatively, if Commerce does not exclude cage data it must explain the anomaly and its conclusion that the IAS data is nonetheless reliable.

Factors of Production R.

Plaintiffs make other challenges to Commerce's findings including challenges to the financial ratios, labor, sawdust, rice husk, fish waste, fish belly, fish skin, frozen broken meat, fresh broken meat, and "freight in" for the selling, general and administrative ("SG&A") expense ratio. For all of these issues. Commerce has either relied in part upon the selection of the primary surrogate country, or if Commerce were to select a different primary surrogate country upon remand it might consider the selection of the primary surrogate country in its analysis. Likewise, Commerce's analysis of the inland freight and brokerage and handling calculation is contingent upon its primary surrogate country selection. Therefore, the court reserves judgment on all issues pertaining to the values of the factors of production in this case until after remand.

C. **Fish Oil Byproduct Offset**

Defendant requests a voluntary remand to reconsider its calculation of Vinh Hoan's byproduct. "Generally, a request for a voluntary remand due to substantial and legitimate agency concerns should be granted." Timken Co. v. United States, 38 CIT __, __, Slip Op. 14-51, *5 (May 2, 2014) (citing SKF USA Inc. v. United States, 254 F.3d 1022, 1029 (Fed. Cir. 2001)). "Commerce's concerns are substantial and legitimate when (1) commerce has a compelling justification for the remand, (2) the justification for remand is not outweighed by the need for finality, and (3) the scope of the remand is appropriate." Timken Co., Slip Op. 14–51 at *5 (citing Ad Hoc Shrimp Trade Action Comm., 37 CIT __, __, 882 F.Supp.2d at 1381).

Defendant justifies its remand request by explaining Commerce "used the methodology to calculate Vinh Hoan's by-product for the first time in the *Final Results*" and it did not have "the opportunity to consider [the parties'] arguments in the first instance" outside the ministerial error context. Def.'s Resp. at 80. Thus, if Defendant were to address the parties' arguments here, it would amount to "impermissible post hoc rationalization." Id. Although Defendant does not concede error, both Vinh Hoan and Catfish Farmers of America have raised several issues with the fish oil calculation and byproduct offset. Here, the justification for remand is not outweighed by the need for finality especially where both Vinh Hoan, a mandatory respondent in the administrative proceeding, and Catfish Farmers of America, a petitioner in the administrative proceeding, have challenged Commerce's fish oil calculation and byproduct offset methodology. See Vinh Hoan's Br. at 7–19; Catfish Farmers of America's Br. at 17–25. Thus, parties on both sides of the proceedings are seeking a remand on the very issue that the Defendant is making its request for voluntary remand. Moreover, no party objected to the Defendant's request in their replies. Defendant's request for voluntary remand is granted.

IV. Consignment Sales

Commerce valued Vinh Hoan's credit expenses and inventory carrying costs for its CCEP sales using facts available because it found at verification that "Vinh Hoan calculated the credit expenses for its CCEP sales in the same manner as CEP sales, despite the fact that the shipping dates for the two types of sales differed."²⁷ Decision Memo 52. Commerce calculated credit expenses for CCEP sales "using an average of the days between the dates of shipment from inventory to the date of payment, calculated from the sales traces observed at verification" and inventory carrying costs for CCEP sales "using the average of the days between the date Vinh Hoan's merchandise cleared customs and the date they of shipment [sic] from warehouse." Id. Vinh Hoan argues that Commerce improperly applied facts available to all sales to a specific customer despite "not all sales to that customer [being] consignment sales." Vinh Hoan's Br. 52.

Vinh Hoan challenged this result in its ministerial error comments, but Commerce declined to make the requested change. *See* Ministerial Error Allegation Memorandum 7, PD 466 at bar code 3136028–01 (May 9, 2013) (*"Ministerial Error Allegation Memorandum"*). Specifically, Commerce explained:

We disagree with Vinh Hoan. As we stated in Comment IX of the *Final Results* I&D Memo, we used facts available to calculate these expenses. Thus, the application of these expenses to one of Vinh Hoan's customers was an intentional methodological decision and, therefore, does not constitute a ministerial error pursuant to section 751(h) of the Act and 19 CFR 351.224(f).

Id. at $7.^{28}$

Commerce does not explain in its *Decision Memo* or its *Ministerial Error Allegation Memorandum* its decision to treat all the sales to one customer as if they were consignment, and thus CCEP sales. In its

²⁷ CCEP sales refer to consignment constructed export price sales, a variation of constructed export price sales made on consignment. Constructed export price "means the price at which the subject merchandise is first sold (or agreed to be sold) in the United States before or after the date of importation by or for the account of the producer or exporter of such merchandise or by a seller affiliated with the producer or exporter, to a purchaser not affiliated with the producer or exporter" 19 U.S.C. § 1677a(b).

²⁸ Commerce's reference to "Comment IX" is meant to refer to Comment XII of the *Decision Memo* where Commerce discussed "Vinh Hoan's Imputed Expenses for Constructed Export Price."

Decision Memo, Commerce merely explains how it treats Vin Hoan's CCEP sales, it does not identify which sales it treated as CCEP sales. Decision Memo 52 (explaining that "[f]or these final results, as facts available, the Department has valued Vinh Hoan's credit expenses for its CCEP sales using an average of the days between the dates of shipment from inventory to the date of payment, calculated from the sales traces observed at verification."). In response to Vinh Hoan's ministerial error comment that Commerce had mistakenly identified all sales to one customer as CCEP sales. Commerce asserted that it made a methodological choice, not a mistake. See Vinh Hoan Ministerial Error Comments at 3-4, CD 266 at bar code 3125191-01 (Mar. 20, 2013); Ministerial Error Allegation Memorandum 7. If this was indeed Commerce's choice, then Commerce needs to explain why its choice was reasonable in light of the fact that there is record evidence that clearly demonstrates that not all of Vin Hoan's sales to the one customer were CCEP sales. The court remands this issue to Commerce to reconsider its decision to treat all sales to one customer as consignment sales or explain why it is doing so despite record evidence that only some sales to such customers were consignment sales.

V. Inclusion of Sample Transactions

Vinh Hoan argues that Commerce improperly included sample transactions with positive gross unit prices in Vinh Hoan's margin calculation. Vinh Hoan Br. 51. Vinh Hoan devotes only three sentences to this argument asking the court to remand the decision because "these transactions should have been excluded from the margin calculation." *Id.* Commerce rejected this argument in its Ministerial Error Allegation Memorandum explaining that it "does not exclude sales from the margin calculation unless a party provides a compelling reason to do so" and here "Vinh Hoan provided the Department with no such reasoning." Ministerial Error Allegation Memorandum at 8 (citation omitted). Commerce's determination was reasonable and is affirmed.

VI. Net vs. Gross Weight Adjustment

Catfish Farmers of America argue that Commerce's refusal to make an adjustment to Vinh Hoan's NV was not supported by substantial evidence and was otherwise not in accordance with law. More specifically, Catfish Farmers of America argue that Commerce's determination was not supported by substantial evidence because it did not adequately explain why it would not adjust Vinh Hoan's FOP data. Additionally, Catfish Farmers of America argue that Commerce's determination was contrary to its past practice in prior reviews. Given the record before the court and Commerce's past practice, Commerce has failed to adequately explain how its refusal to adjust Vinh Hoan's NV is reasonable and makes a fair comparison as required by the statute.

Commerce conducts administrative reviews to "review, and determine . . . the amount of any antidumping duty" by calculating "the normal value and export price . . . of each entry of subject merchandise, and the dumping margin for each such entry." 19 U.S.C. § 1675(a). The AD statute defines the term "dumping margin" as "the amount by which normal value exceeds the export price or constructed export price of the subject merchandise." 19 U.S.C. § 1677(35)(A). "[A] fair comparison shall be made between the export price or constructed export price and normal value." 19 U.S.C. § 1677b(a). As discussed above, in NME cases, NV is determined "on the basis of the value of the factors of production utilized in producing the merchandise" 19 U.S.C. § 1677b(c)(1)(B).

Here, Commerce selected Vinh Hoan as one of two mandatory respondents for the eighth administrative review. *See Preliminary Results* at 56,180. Responding to Department questionnaires, Vinh Hoan explained that it used its reported gross quantity of tra product in kilograms as its denominator to value most of its FOPs.²⁹ *See* Vinh Hoan's Section D QR at 4, 9, CD 59 at bar code 3049386–01 (Jan. 3, 2012). FOPs were then calculated on a US dollar to kilogram basis (USD/kg).

At verification, Commerce discussed glazing with Vinh Hoan company officials. Glazing is a process that adds weight to frozen fish fillets by "coating the finished fish fillet with water and then freezing it." Catfish Farmers of America's Br. 6–7. In its report, Commerce stated

[w]e inquired as to whether the company's net price includes glazing. Company officials indicated that net price does not include glazing. Company officials confirmed that U.S. products did not include glazing during the POR, but other products sold that were included as part of the denominator did include glazing.

See Verification of Vinh Hoan Export Price at 16, CD 249 at bar code 3110869–01 (Dec. 17, 2012). In other words, Vinh Hoan's net sales price did not include glazing, Vinh Hoan's U.S. products price during the POR did not include glazing, however some of the sales that were included in the denominator used to calculate FOPs did include glazed sales. Commerce also conducted a "deglazing test" on one box

²⁹ The gross quantity of tra product refers to Vinh Hoan's total production of glazed and unglazed merchandise sold worldwide, which Commerce required Vinh Hoan to report.

of merchandise. See id. at 45. After the test was completed the officials observed and recorded the difference in weights between the net weight and gross weight of the fish. Id. Vinh Hoan also reported that "the subject merchandise [it] sold [] did not include ice, water, glazing, etc. weight." Vinh Hoan's Section C Questionnaire Response at 9, CD 72 at bar code 3049432-01 (Jan 3, 2012). Vinh Hoan reported the total quantity consumed of a given factor in kilograms for production of all tra products during the POR and divided that number by the total quantity of finished tra products on a kilograms basis. This created a ratio which represented the kilograms consumed of the factor per kilogram of finished product. See, e.g., Vinh Hoan's Third Supplemental Questionnaire Response at Ex. 1, CD 183 at bar code 3090578–01 (Aug. 2, 2012) (showing Vinh Hoan's reported whole fish consumption and ratio).³⁰ In other words, Vin Hoan reported its FOPs using a gross weight denominator (a denominator that included the weight of glazing), and although it sold glazed tra elsewhere, it did not sell any glazed tra in the United States.

Commerce rejected Catfish Farmers of America request for an adjustment to Vinh Hoan's FOP data explaining that "Vinh Hoan's U.S. sales are reported on a gross weight basis . . . [but] [b]ecause none of Vinh Hoan's U.S. customers requested ice, water, or glazing, its reported net weight and gross weight were the same." *Decision Memo* 48. Moreover, Commerce explained, "Vinh Hoan reported . . . that its factors products were reported on a gross weight basis" *Id.* Commerce discussed its analysis from the second administrative review. Finally, it explained that "because the NV and U.S. price were [sic] data were both reported on a gross weight basis, the relevant basis of comparison is consistent; therefore, no adjustment is necessary or warranted." *Id.*

Commerce's determination is not supported by substantial evidence. Catfish Farmers of America, has demonstrated with record evidence below that (1) Vinh Hoan's U.S. sales did not include any glazed sales; and (2) each factor ratio reported by Vinh Hoan was calculated with a denominator that includes glazed weight. Commerce has not provided an adequate explanation for why no adjustment is necessary to make a fair comparison or why it cannot make such an adjustment. No party disputes that for Vinh Hoan's U.S. sales the net and gross weights are equal. No party disputes that Vinh Hoan reported its FOPs on a gross weight basis. The explana-

³⁰ For example, to determine the whole fish input FOP on a USD/kg basis, the gross quantity of whole fish consumed during the POR, in kilograms, was divided by the total aggregate quantity in kilograms of all tra products sold during the POR. This ratio was then multiplied by the surrogate value selected by Commerce for whole fish. This method was repeated for each FOP.

tion Commerce has provided misses the point and does not explain why Commerce should not make an adjustment. Commerce has not credited some facts over others. Instead, Commerce has ignored facts cited by Catfish Farmers of America, and responded to petitioner's arguments by reciting undisputed facts. The substantial evidence standard requires more than this. It requires that Commerce "provide a 'rational connection between the facts found and the choice made . . . [and] articulate a satisfactory explanation for its action."" *Baroque Timber Industries*, 38 CIT __, __, 971 F. Supp. 2d at 1339–40 (citing *Burlington Truck Lines, Inc.*, 371 U.S. at 168; *Yangzhou Bestpak Gifts & Crafts Co.*, 716 F.3d at 1378). Here, Commerce has not done so.

In the original investigation Issues and Decision Memorandum, Commerce explained that for the preliminary results it compared a respondent's U.S. sales on a net weight basis to the factors of production on a gross weight basis. Issues and Decision Memorandum for Certain Frozen Fish Fillets from the Socialist Republic of Vietnam 4, A552-801, (June 16, 2003), available at http://enforcement.trade.gov/ frn/summary/vietnam/03-15794-1.pdf (last visited February 12, 2015) ("Investigation Decision Memo"). After comments from the parties. Commerce decided to make adjustments to the FOPs and to compare the U.S. sales and FOPs both on a net weight basis in a post-preliminary determination. Id. For the final results, Commerce obtained more information and made its comparison on a gross weight to gross weight basis. Id. It explained that its practice is to compare FOPs on the basis that respondents sell their products in the U.S. market and thus, it preferred a comparison on a gross weight basis because that was how the respondent made its U.S. sales. Id. at 5. In the investigation, respondents reported their U.S. prices on both net and gross weight bases, but where the net and gross weights were equal, there was no distinction made between the net and gross weights. Id. at 4 nn.7, 8. This fact indicates that both glazed and unglazed sales were made to the U.S. during the POI. It is unclear if the same rationale used in the investigation would apply to this review where Vinh Hoan has not made any glazed sales to the U.S. market. The U.S. sales were, in effect, reported on both a net weight and gross weight basis because as Commerce has explained, in this review, gross and net weight were the same for the U.S. sales. Commerce must provide a rational explanation on remand for its choice to use a ratio with a denominator that includes glazing weight despite the fact that there were no glazed sales made to the US market.

In its discussion of the second administrative review, Commerce again misses the point. In that review, one of the respondents argued for an upward adjustment of its U.S. price, which had been reported on a net weight basis, by the glazing percentage, so that U.S. price and NV would be compared on the same basis. Commerce rejected the requested adjustment finding that both companies had reported their FOPs on a net weight basis, the same basis on which the respondents had reported their U.S. price. Commerce further explained that although respondents had reported glazing water and electricity as FOPs, no adjustment was required because it did not affect the per unit basis of comparison by making inconsistent denominators.³¹ Here, Commerce explains that the facts are different because the U.S. prices and NVs were both reported on a gross weight basis. Presumably, under Commerce's current rationale, if Vinh Hoan had reported its U.S. sales on a net weight basis, which is the same weight reported here, Commerce would have made an adjustment to the FOP data. Relying on the word used by Vinh Hoan, either net or gross is simply not substantial evidence. Again, here the denominators can only be considered consistent if we ignore the evidence showing, (1) there are no glazed U.S. sales; and (2) there are glazed sales in the FOP denominator. Commerce's explanations in its past investigation and reviews make clear the reason for an adjustment is that inconsistent denominators lead to a distorted picture because Commerce compares a ratio based upon one type of sale, *i.e.*, glazed (also known as a gross sale) when the US sales were actually of another type, *i.e.*, unglazed (also known as a net sale). The relevant question is whether the denominators represent the same type of sale: glazed or unglazed. The fact that sometimes a net weight sale and a gross weight sale would be the same because there were no glazed sales does not alter the fact that there were two different types of sales made. Thus, Commerce must consider this evidence on remand and either make an adjustment that is supported by the record or articulate a rational explanation for why no adjustment is necessary or cannot be made.³²

³¹ The court notes that the circumstances of this case appear to be what Commerce described in the 2nd Administrative Review. The per unit basis of comparison is affected by a denominator for U.S. sales with no glazed weight and a denominator for NV that includes glazed weight.

³² At oral argument, counsel for the government argued that Commerce's path was reasonably discernable. Counsel explained that Commerce did not have evidence which would allow it make an adjustment. Moreover, counsel explained that the lack of evidence in this review makes the review distinguishable from times when Commerce had made an adjustment in the past. While Commerce did note that "Vinh Hoan reported . . . its factors products were on a gross weight basis . . . [and] that all of its production facilities use water in the production process," *Decision Memo* 48, Commerce did not explain that it could not make an adjustment based on the way the information was reported, or why allowing for a distortion to the denominator was more accurate than leaving potential distortions in the numerator. Further, Commerce specifically analogized this review to the second administrative review after explaining that in the second administrative review no adjustment was

CONCLUSION

In accordance with the foregoing, it is hereby

ORDERED that Commerce's decision regarding sample sales is sustained; it is further

ORDERED that Commerce's primary surrogate country selection is remanded for further consideration consistent with this opinion; it is further

ORDERED that Commerce's refusal to adjust NV is remanded for further consideration consistent with this opinion; it is further

ORDERED that Commerce's use of facts available for all sales to one customer as opposed to all CCEP sales is remanded for further consideration consistent with this opinion; it is further,

ORDERED that Commerce's fish oil calculation is remanded in accordance with Commerce's request; it is further,

ORDERED that Commerce shall file its remand determination with the court within 60 days of this date; it is further,

ORDERED that the parties shall have 30 days thereafter to file objections; and it is further,

ORDERED that the Government and Defendant-Intervenors shall have 15 days thereafter to file responses.

Dated: February 19, 2015

New York, New York

/s/ Claire R. Kelly Claire R. Kelly, Judge

Slip Op. 15–20

UNITED STATES, Plaintiff, v. American Casualty Co. of Reading, Pennsylvania, Rupari Food Services, Inc., and William Vincent Stilwell, A/K/A "Rick" Stilwell, Defendants.

> Before: Nicholas Tsoucalas, Senior Judge Consol. Court No.: 10-00119

[Defendants' Joint Motion to Strike Plaintiff's Motion for Summary Judgment is denied. Defendants' Joint Motion to Stay Summary Judgment until the Motion to Dismiss is resolved is granted. Defendants' Consent Motion for an Extension of Time to Reply to Plaintiff's Opposition is granted.]

Dated: March 4, 2015

Mikki Cottet, Senior Trial Counsel, Commercial Litigation Branch Civil Division, U.S. Department of Justice, of Washington, DC, for Plaintiff. With her on the brief were *Joyce R. Branda*, Acting Assistant Attorney General, *Jeanne E. Davidson*, Director, and

necessary because the per unit basis of comparison was not based on inconsistent denominators. Here the denominators are inconsistent. The court cannot say that Commerce's path was reasonably discernable.

Patricia M. McCarthy, Assistant Director. Of counsel on the brief was Brian J. Redar, U.S. Customs and Border Protection Office of Associate Chief Counsel, Long Beach, CA.

Peter A. Quinter, Gray Robinson, P.A., of Miami, FL, and Lawrence M. Friedman, Barnes Richardson & Colburn, of Chicago, IL, for Defendants Rupari Food Services, Inc. and William Vincent Stilwell a/k/a "Rick" Stilwell.

Frances P. Hadfield, Grunfeld, Desiderio, Lebowitz, Silverman & Klestadt, LLP, for Defendant American Casualty Co. of Reading, Pennsylvania.

OPINION AND ORDER

Tsoucalas, Senior Judge:

The issue before the court is whether to grant American Casualty Co. of Reading Pennsylvania, Rupari Food Services Inc., and William Vincent Stilwell a/k/a "Rick" Stilwell's (collectively "Defendants") Motion to Strike Plaintiff's Motion for Summary Judgment and Opposition to Motion to Dismiss; or in the alternative, stay the Motion for Summary Judgment until Defendants' Motion to Dismiss is resolved.

On October 27, 2014, this court ordered the following: that Defendants file a Motion to Substitute a Revised Motion to Dismiss within seven days of the order; that Plaintiff's response shall be filed on or before January 15, 2015; and "any additional dispositive motions shall be filed within 45 days after a decision denying in whole or in part Defendants' motion to dismiss" Order at 1–2, October 27, 2014. ECF No. 71.

Defendants filed their revised Motion to Dismiss on November 3, 2014. Defs.' Mot. to Dismiss, November 3, 2014, ECF No. 76. On January 15, 2015, Plaintiff filed a Motion for Summary Judgment and Opposition to the Motion to Dismiss. Pl.'s Mot. For Summ. J. and Opp'n to Mot. to Dismiss, January 15, 2015, ECF No. 79.

On January 22, 2015, Defendants filed a Joint Motion to Strike Plaintiff's Motion for Summary Judgment, or in the alternative, to stay further action on the Motion for Summary Judgment until the Motion to Dismiss is resolved. Defs.' Joint Mot. to Strike Pl.'s Mot. for Summ. J. and Consent Mot. for an Extension of Time, January 22, 2015, ECF No. 82 ("Defs.' Br."). Additionally, the parties consented to an extension of time for the Defendants to reply to Plaintiff's Opposition to Defendants' Motion to Dismiss. Id.

"The granting of a motion to strike constitutes an extraordinary remedy, and should be granted only in cases where there has been a flagrant disregard of the rules of court." Jimlar Corp. v. United States, 10 CIT 671, 673, 647 F. Supp. 932, 934 (1986). Motions to strike are a drastic remedy that are not favored and infrequently granted. Ta Chen Stainless Steel Pipe Co. v. United States, 31 CIT 794, 810 (2007) (not reported in Federal Supplement). Accordingly, "courts will not grant motions to strike unless the brief demonstrates a lack of good faith, or that the court would be prejudiced or misled by the inclusion in the brief of the improper material." *Jimlar Corp.*, 10 CIT at 673.

The Defendants argue that the court should strike Plaintiff's Motion for Summary Judgment, because the motion has "referenced new information not referenced in the pleadings and not previously disclosed to the parties for which the authenticity and factual accuracy have not been determined." Defs.' Br. at 6. The Defendants contend that such information would prejudice the court. *Id.* The Defendants point to two specific examples of this in Plaintiff's Motion for Summary Judgment: (1) the Declaration of Dr. Greg Lutz, to support the allegation that the crawfish did not originate in Thailand; (2) the Declaration of Richard Porter which references an unrecorded conversation with a Rupari salesperson regarding the origin of the crawfish. *Id.*

First, with respect to Dr. Greg Lutz, Defendants claim that his identity was not disclosed during discovery. *Id.* Plaintiff counters that it was not required to disclose Dr. Greg Lutz's identity to the Defendants, and the Declaration is consistent with other record evidence before the court which remains unchallenged by Defendants. Resp. to Defs.' Mot. to Strike at 6–7, February 10, 2015, ECF No. 89 ("Pl.'s Br.").

It appears from the exhibits submitted by Plaintiff that Dr. Greg Lutz's identity was not disclosed during discovery, however, USCIT Rule 26(a)(2)(c) requires the disclosure of experts ninety days before the date set for trial, unless ordered otherwise by the Court. Id. at Ex. 1, 2; USCIT R. 26 (a)(2)(c). Here, the court has not set a date for trial and Defendants have not shown that the court ordered disclosure of Dr. Greg Lutz's identity. Defs.' Br. at 1–9. Thus, Plaintiff was under no obligation to disclose Dr. Greg Lutz's identity. USCIT R. 26 (a)(2)(c). Nevertheless, Defendants should have the opportunity to depose Dr. Greg Lutz and review his credentials and publications before having to respond to Plaintiff's Motion for Summary Judgment. See Baron Services Inc. v. Media Weather Innovations LLC, 717 F.3d 907, 908-913 (Fed. Cir. 2013) (vacating summary judgment order as premature, where Plaintiff did not have an opportunity to depose two witnesses who provided affidavits in support of a motion for summary iudgment).

Second, Defendant, American Casualty, argues that Plaintiff did not previously disclose Mr. Richard Porter's identity or the contents of his Declaration. Defs.' Br. at 6. Moreover, Defendants collectively claim that the Declaration was neither mentioned in the Complaint, nor related to any factual allegation in the Complaint. *Id.* As such, Defendants believe that this Declaration will prejudice the court. *Id.* In contrast, Plaintiff insists that it disclosed Mr. Richard Porter's identity to all Defendants twice. Pl.'s Br. at 7.

The court rejects Defendants' arguments. Upon examination of the discovery materials, the court finds that Plaintiff disclosed the identity of Mr. Richard Porter to all Defendants in its Amended Initial Disclosures on March 25, 2013, and in its Second Amended Initial disclosures on July 25, 2013. Pl.'s Br. at Ex. 1, 2. Additionally, Defendants, in their Consent Motion to Amend the Scheduling Order, explicitly referred to Mr. Richard Porter. Consent Mot. to Amend Scheduling Order at 2, April 12, 2013, ECF No. 32. This reference further corroborates that Defendants knew Mr. Richard Porter's identity. See id.

In deciding a motion to dismiss, a Court may consider documents not attached to the complaint where the authenticity of those documents is not in question, and the factual allegations of the complaint revolve around the documents. *Young v. Lepone*, 305 F.3d 1, 11 (1st Cir. 2002). Here, although Mr. Richard Porter's Declaration was not attached to the Complaint, the Declaration is related to the factual allegation in the Complaint that Rupari and its employees knew that the crawfish tail meat did not originate in Thailand. Compl. at ¶30, June 20, 2011, ECF No. 2. Accordingly, the court may consider the Declaration. *See Young*, 305 F.3d at 11.

Finally, Defendants argue that Plaintiff's Motion for Summary Judgment is premature based upon the court's Scheduling Order of October 27, 2014. Defs.' Br. at 4. The court agrees. The court's Scheduling Order of October 27, 2014, requires that the court rule upon the Motion to Dismiss before any other dispositive motions are filed by either party. Order at 1–2. Furthermore, the order specifically states that any additional dispositive motions should be filed within 45 days *after* a decision denying in whole or in part Defendants' Motion to Dismiss. *Id.* (Emphasis added). On this basis alone, Plaintiff's Motion for Summary Judgment is premature and consequently improper. *See id.*

Notwithstanding Plaintiff's improper filing of the Motion for Summary Judgment, the court declines to strike Plaintiff's brief in its entirety, because Plaintiff properly submitted its Opposition per the court's October 27, 2014 order. *See Jimlar Corp.*, 10 CIT at 673 (denying motion to strike and finding that motions to strike are a "drastic remedy.")

Accordingly, Defendants' Motion to Strike is denied. Defendants' Motion to Stay Further Action on Plaintiff's Motion for Summary Judgment until the court issues a decision on the Motion to Dismiss is granted. No other dispositive motions will be considered until the Motion to Dismiss is ruled upon. As such, Plaintiff shall refile its Opposition without the Motion for Summary Judgment within ten days of this Order. Defendants shall have fourteen days from the date Plaintiff resubmits its Opposition to file a Reply. Plaintiff shall have fourteen days, if necessary, from the date the Motion to Dismiss is denied, in whole or in part, to file an Amended Motion for Summary Judgment. Defendants shall have forty-five days from the expiration of the fourteen day period to respond to Plaintiff's Motion for Summary Judgment, and if necessary, file a Cross-Motion for Summary Judgment.

ORDER

Upon consideration of Defendants' Motion to Strike or in the Alternative Stay Further Action on Plaintiff's Motion for Summary Judgment until the Motion to Dismiss is resolved, the response to this motion and the papers and proceedings herein, it is hereby

ORDERED that Defendants' Motion to Strike is **DENIED**; it is further

ORDERED that Defendants' Motion to Stay Further Action on Plaintiff's Motion for Summary Judgment until the court issues a decision on the Motion to Dismiss is **GRANTED**; it is further

ORDERED that no other dispositive motions will be considered until the Motion to Dismiss is ruled upon; it is further

ORDERED that Plaintiff shall refile its Opposition to the Motion to Dismiss without the Motion for Summary Judgment within ten days of this order; it is further

ORDERED that Defendants shall have fourteen days from the date Plaintiff resubmits its Opposition to file a Reply; it is further

ORDERED that Plaintiff shall have fourteen days, if necessary, from the date the Motion to Dismiss is denied, in whole or in part, to file an Amended Motion for Summary Judgment; it is further

ORDERED that Defendants shall have forty-five days from the expiration of the fourteen day period to respond to Plaintiff's Motion for Summary Judgment, and if necessary, file a Cross-Motion for Summary Judgment.

SO ORDERED.

Dated: March 4, 2015 New York, New York

> Nicholas Tsoucalas Nicholas Tsoucalas Senior Judge